IEEEAccess Multidisciplinary : Rapid Review : Open Access Journal

Received 27 April 2023, accepted 11 May 2023, date of publication 15 May 2023, date of current version 24 May 2023.

Digital Object Identifier 10.1109/ACCESS.2023.3276351

RESEARCH ARTICLE

Generalized Defensive Modeling of Malware Propagation in WSNs Using Atangana–Baleanu– Caputo (ABC) Fractional Derivative

VINEET SRIVASTAVA^{®1}, PRAMOD KUMAR SRIVASTAVA^{®1}, JYOTI MISHRA¹, RUDRA PRATAP OJHA^{®2}, PURNENDU SHEKHAR PANDEY^{®3}, (Senior Member, IEEE), RADHE SHYAM DWIVEDI^{®1}, LORENZO CARNEVALE^{®4}, (Member, IEEE),

AND ANTONINO GALLETTA^{[0,4,5,6}, (Member, IEEE) ¹Rajkiya Engineering College Azamgarh, Azamgarh, Uttar Pradesh 276201, India

²Department of Computer Science and Engineering, GL Bajaj Institute of Technology and Management (GLBITM), Affiliated to AKTU, Greater Noida, Uttar Pradesh 201306, India

³Department of Electronics and Communication Engineering, GL Bajaj Institute of Technology and Management (GLBITM), Affiliated to AKTU, Greater Noida, Uttar Pradesh 201306, India

⁴MIFT Department, University of Messina, 98166 Messina, Italy

⁵INDAM Gruppo Nazionale per il Calcolo Scientifico (GNCS), 44122 Ferrara, Italy

⁶CINI, Laboratorio Informatica e Società (IeS), 00198 Lazio, Italy

Corresponding author: Antonino Galletta (angalletta@unime.it)

This work was supported in part by the Department for Drug Policies of the Presidency of the Council of Ministers (Italian Government) through the Project InstradaME under Grant CUP F49I20000100001, and in part by the Italian Ministry of Health Piano Operativo Salute (POS) Trajectory 2 "eHealth, diagnostica avanzata, medical device e mini invasività" through the Project "Rete eHealth: AI e strumenti ICT Innovativi orientati alla Diagnostica Digitale (RAIDD)" under Grant CUP J43C22000380001.

ABSTRACT The malware spreading in Wireless Sensor Network (WSN) has lately attracted the attention of many researchers as a hot problem in nonlinear systems. WSN is a collection of sensor nodes that communicate with each other wirelessly. These nodes are linked in a decentralised and distributed structure, allowing for efficient data collection and communication. Due to their decentralised architecture and limited resources, WSN is vulnerable to security risks, including malware attacks. Malware can attack sensor nodes, causing them to malfunction and consume more energy. These attacks can spread from one infected node to others in the network, making it essential to protect WSN against malware attacks. In this paper, we focus on the analysis of a novel fractional epidemiology model, specifically the fractional order SEIVR epidemic model in the sense of Caputo's fractional derivative of order $0 < \alpha \le 1$ with the goal of examining the efficacy of vaccination strategies and the heterogeneity of a scale-free network on epidemic spreading. First, using the next-generation technique and obtain the basic reproduction number of the proposed epidemic model, which is essential for determining both the locally asymptotically stable equilibrium point of the wormfree system and the unique existence of the endemic equilibrium point. To numerically solve the model, the Adam-Bashforth-Moulton predictor-corrector (ABM) method is applied. The fractional calculus enables us to deal directly with the "memory effect" of numerous phenomena, taking into account the system's dependence on previous stages. This method provides the results of a complex system. Additionally, research demonstrates that vaccine treatments are quite effective at preventing the spread of malware. The outcome of the study reveals that the applied ABM predictor-corrector method is computationally strong and effective to analyse fractional order dynamical systems in the SEIVR epidemic model for malware propagation in WSN. The results show that the order of the fractional derivative has a significant effect on the dynamic process. Also, from the result, it is obvious that the memory effect is zero for $\alpha = 1$. When the fractional order α is decreased from 1, the memory effect appears, and its dynamics vary according to the time. This memory effect points out the difference between derivatives of fractional and integer orders. The theorems and their proofs are presented to validate the validity of the proposed model. To validate the proposed model, extensive theoretical study and computational analysis have also been applied.

The associate editor coordinating the review of this manuscript and approving it for publication was Stefano Scanzio.

INDEX TERMS Basic reproduction number, stability analysis, fractional order derivative.

I. INTRODUCTION

Wireless Sensor Network (WSN) is a type of network where a large number of small sensor nodes are deployed in a specific area to collect environmental data and transmit it to a central node or sink. The applications of WSN are very wide including defense, agriculture, healthcare, defense, mobility, commerce, industry, personal use and more [1], [2]. One of the major challenges faced by WSN is the limited resources available to sensor nodes, including processing power, energy source, memory, and coverage area. As a result, WSN needs to be designed in a way that optimizes these resources and maximizes the network's lifetime.

Another challenge faced by WSNs is security. The sensor nodes are vulnerable to malware attacks, and the data transmitted by the nodes can be intercepted or manipulated. This can lead to a range of problems, including privacy breaches, data tampering, and disruption of network operations.

To address these challenges, researchers have developed various techniques and protocols for optimizing resource usage, improving network security, and extending the lifetime of WSN. These include energy-efficient routing algorithms, data aggregation techniques, and secure communication protocols.

WSNs also have a role to play in the emerging fields of Internet of Things (IoT) and the Internet of Vehicles (IoV) [3], where sensor nodes can be used to collect data from a wide range of sources [4], [5], including vehicles, wearable devices, and smart home appliances. However, the challenges faced by WSNs in these contexts are even greater, as the scale of the networks is much larger, and the data collected is more diverse and complex. Overall, WSNs have enormous potential in a range of applications, but their success depends on addressing the operational challenges faced by these networks such as limited energy, security especially malware attack, and the placement of nodes [6], [7] etc.

One of the most important issues is the malware attacks on WSN. Malware that is created purposely and deliberately installed does not normally enter into a network without human interaction, malware may spread across WSNs since it can be autonomously activated and do so by selftransmission [8]. Wireless device-targeting malicious malware has already been discovered. Cabir is a such type of virus that can spread via the air interface, and is seriously damage the wireless devices.

Researchers have developed various security mechanisms to protect WSNs against malware attacks [9], [10], [11], [12], [13], [14].These include secure routing protocols, intrusion detection systems, and data encryption techniques. However, the nature of WSNs makes it challenging to implement these security mechanisms effectively, given the limited resources available to the sensor nodes.

To address this challenge, researchers have turned to mathematical modelling to better understand the dynamics of malware propagation in WSNs. Epidemiological models, which are commonly used to study the spread of diseases in populations, can be adapted to study the spread of malware in WSNs. By developing mathematical models of propagation, researchers can better understand and monitor the patterns of malware transfer from one node to the next. The analysis of disease dynamics continues to be a hot topic among scientists [15], [16], [17]. The transmission of malware in WSN is similar to the spread of disease in populations. In both cases, the spread of the contagion depends on various factors such as the network topology, connectivity, and transmission rates. The various epidemiological models have been proposed by researchers to investigate that stability and performance of WSN [18], [19], [20], [21], [22], [23].

The ordinary differential equation (ODE) is used to describe the various physical phenomena of the model. The ODE helps in the formulation of the models and analyses the dynamics of malware spreading in the WSN. But classical calculus is not able to describe the exact complex phenomena of the system. To describe the exact complex phenomena of the system, use fractional calculus (FC). Fractional calculus extends the concepts of differentiation and integration to noninteger orders. Fractional differential equations (FDEs) have been shown to model many physical phenomena more accurately than ODEs, particularly in systems that exhibit memory effects, or long-range dependence. One of the key advantages of FDEs is that they provide a more accurate description of the behaviour of complex systems than ODEs, which assume instantaneous and proportional responses. FDEs can capture the dynamics of systems that have a complex history or memory, where past events can influence the system's behaviour [24].

To replicate practical problems, fractional derivatives like Caputo, Grünwald Letnikov, Riemann-Liouville, Jumarie etc. have gained popularity among scholars. These derivatives' theoretical underpinnings have advanced greatly over time [25], [26]. Some applications of the FC have been explained by the researchers [27], [28], [29].

In the case of dynamical systems, the incorporation of the fractional derivative is significant since the definition includes integration, and as a result, the function stores information regarding the past memory. The study by Bolton et al. [30] shown the superiority of fractional-order models versus integer-order models in terms of their applicability to a given data set for analysis. Many researchers [31], [32], [33], [34] use applications of these derivatives in their work to comprehend many phenomena in mathematical biology and their multidisciplinary disciplines. In this context, different researchers have also begun investigating the classical order epidemiological epidemic models by integrating a wide variety of fractional derivatives [35], [36], [37].

In spite of all above said applications, both the derivative in the Riemann-Liouville sense and the derivative in the Caputo sense have some drawbacks, as has been noted. (1) There was no nonlocality in the kernel. (2) The integral associate is the average of the function and its integral, not a fractional operator. Some academics draw the conclusion that the operator was a fractional parameter filter and not a derivative with fractional order. Consequently, the fractional parameter may be seen as a filter regulator. The fact that their kernels are solitary, although being nonlocal, poses a significant challenge for the well-known Caputo and Riemann-Liouville orders. When simulating situations in the actual world, this flaw has an impact

Recently, new forms of nonlocal fractional derivatives have been proposed in the literature to handle the powerlaw reduction of derivative operators. The Caputo-Fabrizio derivative with fractional order was the new operator presented by Caputo and Fabrizio [38]. Numerous researchers successfully used their derivative in a small number of realworld issues as a result of the originality of their findings [39], [40]. Their operator is new in that the derivative, which has applicability in several groundwater and thermal science issues, lacks a single kernel. In addition to the innovative idea's real-world implementations, several theoretical papers were also presented.

Atangana and Baleanu proposed a novel operator based on the Mittag-Leffler function with fractional order to address the aforementioned issues [41]. In addition to using a nonlocal kernel, its operators have all the advantages of Caputo and Fabrizio. All the advantages of the Riemann-Liouville and Caputo operators are present in the operators, and the kernel is non-singular. It's interesting to note that their fractional integral is the fractional average of the supplied function's Riemann-Liouville fractional integral and the function itself. Along with the aforementioned advantages, the derivative was discovered to be quite helpful in thermal science and material sciences [41], [42]. These novel derivatives with fractional orders are both fractional derivatives and filters. The Atangana-Baleanu-Caputo (ABC) fractional derivative provides an accurate description of the memory [43]. The important applications of the ABC operator can be found in [44], [45], [46], [47], [48], [49], [50], [51], [52], [53], [54], [55], [56], [57], and [58]. Recently, it has been observed that fractional differential equations can be employed to modeling phenomena of worldwide more accurately. The global problem of the spread of the disease attracted the attention of researchers from various fields, which led to the emergence of a number of proposals to analyze and anticipate the development of the epidemic [53].

To understand the attacking and spreading dynamics of malware in WSN, Author [59] proposed a compartmental epidemic model to examined the projected non-linear structure of the model by using generalized Adams-Bashforth-Moultan method.

The Adams-Bashforth has been recognized as a great and powerful numerical method able to provide a numerical solution closer to the exact solution [60]. The correct version of the fractional Adams-Bashforth [61] methods which take into account the nonlinearity of the kernels including Mittag-Leffler law for the Atangana-Baleanu Differential operator, which is used here to model WSN.

The vaccination technique is not entirely effective because vaccinations may only provide partial protection from infection after a particular period of time. To address this problem, we propose a fractional epidemic (SEIRV) model with a factor $\rho \in [0, 1]$ in the contact rate of vaccinated nodes, where $\rho = 0$ means perfect efficacy of vaccination and $\rho = 1$ means complete inefficacy of vaccination. The FDEs will help in the exact analysis of vaccination effect on WSN.

The key goal of proposed models is to develop method to control the malware transmission as well as to enhance WSN lifetime. The model describes the transmission of malware dynamics in WSN, the contributions of proposed model are:

1. The proposed model studies the transmission dynamics of worms in WSN. The model suggests the mechanism against malware attack.

2. For deterrence of malware transmission and consumption of sensor node's energy in WSN, utilize the concept of Exposed state.

3. To include the idea of a vaccination strategy in the epidemic model. This idea is used for the maintenance of network operations. The vaccination method is used to increase network lifetime, suppress malicious activity, and improve WSN security.

4. To analyses the system's response to a malware assault and look into quick WSN recovery techniques under steady state conditions.

5. To investigate the stability of the system under various conditions and to validate the analytical analysis through the results of simulations.

The following order determines how the remaining portion of the paper is structured. Numerical analysis and graphical outcome of the study is presented in the section of results and discussion. In Section II presented related work. Section III presents basic definitions and results. Section IV provides details on the description of the proposed model and the analysis's assumptions. The presence of non-negative solutions and their uniqueness are covered in Section V. Section VI focuses on equilibrium points and their stability analysis. The numerical solution and its analysis are employed in Section VII. Section VIII contains the findings and a discussion of the simulation results for the proposed models. The conclusion and future research are discussed in section IX.

II. RELATED WORK

This section presents a comprehensive review of wireless sensor networks. The number of researchers who have studies spreading of malware in WSN apply the concept of epidemic modeling.

Wu et al. [62] considered a STSIR model to analyse the virus spread among the devices with consideration of variable number of devices. They applied the idea of game theory to curb the virus spread among the devices. Zhang et al. [63] proposed a epidemic theory based model to supress the

spreading of malware in WSNs. They consider heterogeneity in the sensor network as sensor nodes as well as sink nodes. They analysed the effect of malware attack on different types of topologies. They suggest the mechanism of the design, distribution, and maintenance of the sensor network. Further, Ye et al. [64] proposed a SIR1R2 model that discussed about the mechanism of secondary immunity. To analyse the spreading behaviour of virus in WSN computed the value of basic reproduction number (R₀) and found that if R₀ < 1 the malware will no longer survive in WSN, whereas if R₀ > 1 malware will spread in the entire WSN. They studied the impact of secondary immunity on propagation of malware in WSN.

Shen et al. [65] proposed VCQPS model that uses to defend the WSN against the attack of malware. They considered both heterogeneous types of sensor nodes and mobile nodes in the investigation of malware propagation in WSN. To study the dynamics of malware propagation in WSN the differential equations is derived. The changes among the transition states presented by the differential equations. The existence of stationary points of the model is discussed. They compare the proposed model with the traditional models based on epidemic theory and suggest the method to defend WSN against malware attack. Zhou et al. [66] used game theory to analyse the propagation dynamics of malware in WSN. The Nash equilibrium strategy is applied to investigate the propagation behaviour of malware in WSN. The relationship established between ratio of infection steady-state with parameters of the game and discussed the existence of malwares in WSN. They also used cellular automaton and simulate the process of malwares' propagation in WSN and verify the theoretical model correctness with simulation results.

Zhang et al. [67] proposed a MDBCA model which describes the diffusion of malware through use of cellular automata. The set of differential equation deduced which explains the dynamics of various states of the model. The points of equilibria of the model are obtained which determine the threshold value whether malware will diffuse or die out in WSN. They also compute the value of basic reproduction number and investigate the equilibrium points' stability through use of next-generation matrix. The simulation has been performed to validate the effectiveness of the model. A SNIRD model is suggested by Shen et al. [68] to supress spreading of malware in WSN. The suggested model considered the connectivity with heterogeneity of sensor nodes. The model also discussed the hiding characteristics of malware and dysfunctional types of sensor nodes. The existence of equilibrium for the suggested model has been verified, and the threshold value is computed which determines the status of malware in the network, the computed value determine that whether malware will fade out or spread. The SNIRD model compared with the conventional SIS and SIR and demonstrates the effectiveness over SIS and SIR models. Further, for Heterogeneous wireless sensor networks (HWSNs) a HSIRD model is proposed by Shen et al. [69] to analyse the malware propagation dynamics. The model categorise the sensor nodes are of different types. They considered that the sensor nodes may be damage due to attack of malware or physical attack and in both cases sensor nodes will loss the functionality. The connectivity and heterogeneity are considered in the analysis of the malware spread in the system. The points of existence of equilibria of the HSIRD model is discussed. The value of the basic reproduction number is computed which govern the equilibrium points stability. The condition of malware or die out or diffuse is obtained. The model is validated through simulation results. The HSIRD model compared with the traditional SIR and SIS models and showed efficiency over them.

However, there are still some WSN malware spreading issues need to be addressed. The SEIVR model and fractional calculus approach can be used to address the challenges of malware propagation in WSNs.

III. BASIC DEFINITION AND RESULTS

First of all, we recall some basic definitions and results of the fractional calculus.

Definition 1: Let f be a function defined on [a, b] and $\alpha > 0$. The Riemann – Liouville fractional integral of order $\alpha > 0$ for the function f is defined by

$${}^{RL}_{a}D_{x}^{-\alpha}f(x) = \frac{1}{\Gamma(\alpha)}\int_{a}^{x} (x-s)^{\alpha-1}f(s)ds, \ x \in [a,b].$$

Provided the right – hand side is point wise defined on [a, b]. where $\Gamma(.)$ is the gamma function. For $\alpha = 0$, we get ${}_{a}D_{x}^{0} = I$, the identity operator.

Definition 2: Let $\alpha \ge 0$, $n \in \mathbb{N}$ and $f \in C^n[a, b]$. The Caputo fractional derivatives of order α for the function f is defined by

$${}^{C}_{a}D^{\alpha}_{t}f(t) = \frac{1}{\Gamma(1-\alpha)} \int_{a}^{t} \frac{f'(s)}{(t-s)^{\alpha}} ds,$$
$$t \in [a,b], \ 0 < \alpha \le 1$$

Definition 3: Let $\alpha > 0$, the function E_{α} defined by

$$E_{\alpha}(z) = \sum_{n=0}^{\infty} \frac{z^n}{\Gamma(n\alpha+1)} \,, \quad \alpha > 0, \ z \in \mathbb{C}$$

is called the Mittage – Leffler function of order α . The function $E_{\alpha}(z)$ is entire function. For the special cases when $\alpha = 1$, $E_1(z) = e^z$ and when $\alpha = 2$, $E_2(z) = \cosh(\sqrt{z})$. It's general form

$$E_{\alpha,\beta}(z) = \sum_{n=0}^{\infty} \frac{z^n}{\Gamma(n\alpha + \beta)}, \quad \alpha, \beta \in C,$$
$$z \in \mathbb{C} \text{ and } R(\alpha) > 0, \ R(\beta) > 0$$

Definition 4: Caputo-Fabrizio fractional order derivative

$${}^{CF}_{a}D^{\alpha}_{t}f(t) = \frac{B(\alpha)}{1-\alpha} \int_{a}^{t} f'(x) \exp\left(-\frac{\alpha(t-x)}{1-\alpha}\right) dx,$$
$$t \in [a, b], \ 0 < \alpha \le 1$$

where $B(\alpha)$ is the normalization function such that B(1) = B(0) = 1.

Definition 5: Atangana–Baleanu fractional derivative in Caputo sense:

$${}^{ABC}_{a}D^{\alpha}_{t}f(t) = \frac{B(\alpha)}{1-\alpha} \int_{a}^{t} f'(x) \operatorname{E}_{\alpha}\left(-\frac{\alpha(t-x)^{\alpha}}{1-\alpha}\right) dx,$$
$$t \in [a, b], \ 0 < \alpha \le 1$$

Lemma 1: If f is continuous and $\alpha \ge 0$ then ${}_{a}^{c}D_{t}^{\alpha}aD_{t}^{-\alpha}f(t) = f(t)$.

Lemma 2: If $\alpha \geq 0$, r > 0, $\theta \in [-\pi, \pi] \geq 0$, r > 0, $\theta \in [-\pi, \pi]$ and

$$\lim_{t \to \infty} E_{\alpha} \left(-\lambda t^{\alpha} \right) = 0 \text{ for } |\theta| < \frac{\alpha \pi}{2} \lambda = r e^{-i\theta}$$

Now, some result related to the stability of the fractional order systems. Let D be an open subset of \mathbb{R}^n . For $\alpha \in (0, 1]$, we consider the initial value problem (IVP) consisting of an autonomous fractional – order system.

$${}_{a}^{c}D_{t}^{\alpha}u(t) = h(u), \quad u(t_{0}) = u_{0}$$
(2.1)

where $h: D \to \mathbb{R}^n$ is locally Lipschitz continuous in t.

Definition 6: A point u^* is called an equilibrium of the system (2.1) if $h(u^*) = 0$.

Lemma 3: [36], [49] Let $J(u^*)$ denote the cobian matrix of the system evaluated at an equilibrium point u^* , and let λ_j , $j = 1, 2, 3, \ldots, n$ be the eigen values of $J(u^*)$. Then u^* is locally asymptotically stable if and only if $|arg\lambda_j| > \frac{\alpha\pi}{2}$, $j = 1, 2, 3, \ldots, n$.

To prove stability result, we have following lemma:

Lemma 4: Assume that $\alpha \in (0, 1]$ and $g \in C([a, \infty), \mathbb{R}^+)$. For any $\geq a$, we have

$${}_{a}^{c}D_{t}^{\alpha}g^{2}(t) \leq g(t)_{a}^{c}D_{t}^{\alpha}g(t)$$
 [[50], Lemma 1]

$${}_{a}^{c}D_{t}^{\alpha}[g(t) - g^{*} - g^{*}\ln\frac{g(t)}{g^{*}}] \le (1 - \frac{g^{*}}{g(t)})_{a}^{c}D_{t}^{\alpha}g(t),$$

where $g^* \in \mathbb{R}^+$.

Let $H \in C^1(D, \mathbb{R})$ and $\alpha \in (0, 1]$, the α^{th} order Caputo derivatives of H(u) along the solution u(t) of the system ${}_a^c D_t^{\alpha} u(t) = h(u), t \in [a, \infty)$ is given by [65]

$${}_{a}^{c}D_{t}^{\alpha}H(u(t)) = {}_{a}D_{t}^{-(1-\alpha)}\left(\frac{dH}{du}\frac{du}{dt}\right)$$

Now, next is, fractional version of the well-known LaSalle's invariance Principle:

Lemma 5: [37] Assume that A is a bounded closed set in \mathbb{R}^n and that every solution of ${}_a^c D_t^{\alpha} u(t) = h(u), t \in [a, \infty)$,



FIGURE 1. Transition state diagram of the model.

starting from a point in A, remains in A for all time t. Assume, further, that $H \in C'(A, \mathbb{R})$ such that ${}_{a}^{c}D_{t}^{\alpha}H(u(t)) \leq 0$, where u(t) is any solution of the system ${}_{a}^{c}D_{t}^{\alpha}u(t) = h(u)$. Let $E = \{u \in A : {}_{a}^{c}D_{t}^{\alpha}H = 0\}$, and let S be the largest invariant subset of E. Then every solution u(t) of ${}_{a}^{c}D_{t}^{\alpha}u(t) = h(u), t \in$ $[a, \infty)$; originating in A tends to S as $t \to \infty$. In particular, if $S = \{0\}, u(t) \to 0$ as $t \to \infty$.

Theorem 1 [60], [61]: Let f be a continuous function on [a, b]. then, the following inequality is satisfied on a closed interval [a, b]:

$$\left\| {}^{ABC}_{0} D^{\alpha}_{t}[f(t)] \right\| < \frac{M(\alpha)}{1-\alpha} B, \ \|g(t)\| = \max_{a \le t \le b} |g(t)|$$

Theorem 2 [61]: The Atangana-Baleanu fractional derivative in Caputo sense satisfy the Lipschitz condition, for the given functions f and g, the following inequalities can be obtained:

$$\left\| {{^{ABC}_0}D_t^\alpha \left[{f(t)} \right] - {^{ABC}_0}D_t^\alpha [g(t)]} \right\| < H \ \left\| {f(t) - g(t)} \right\|$$

Theorem 3 ([61] Condition for Stability): Let u(t) be a solution of ${}^{ABC}_{0}D^{\alpha}_{t}u(t) = f(t, u(t))$ with f being continuous and bounded;

if f satisfies a Lipschit condition, then the required condition for Adams – Bashforth method when applied to approximate the Atangana – Baleanu derivative of fractional order in Caputo sense is achieved if

$$||f(t_n, u_n) - f(t_{n-1}, u_{n-1})||_{\infty} \to 0 \text{ as } n \to \infty.$$

IV. MODEL DESCRIPTION AND ASSUMPTIONS MADE IN ITS ANALYSIS

The defense mechanism is required for protection of WSN against worm attack. For this purpose, fractional epidemic model SEIVR (Susceptible-Exposed-Infectious-Vaccinated-Recovered) is proposed. The model helps in the study of worm propagation dynamics in WSN, which formulation is described in Fig. 1. There are two types of infectious state of the sensor nodes are considered in the model. These states are: Exposed state sensor node and Infectious state sensor node. Worms transmit in WSN with useful data through the neighboring active mode sensor nodes.

The different states of the proposed models:

Susceptible State(S): The nodes that are not infected but are vulnerable to worms are referred to as susceptible states.

Exposed State(E):The exposed sate contains the infected but non-infectious nodes. This type of nodes does not spread infection to other nodes.

Infectious State(I):Infected nodes are those that have the potential to infect other nodes; these nodes belong to the infected state.

Recovered State(R):When a node that was previously infected is now free of infection, it is considered to be in a state of recovery.

Vaccinated State(V): Those nodes that are vaccinated are considered to be part of the vaccinated state.

For formulation of model the following assumptions has been made:

• Initially sensor nodes are of Susceptible state (S) and they are vulnerable against malware attack. When malware installs in susceptible sensor node then susceptible sensor node becomes exposed node. Malware executed partially in exposed node. The rate of conversion of susceptible node into exposed node is β . Due to normal operation some of susceptible sensor nodes become dead with rate δ_1 .

• The abnormal behavior exhibits y the exposed state (E) of sensor node in WSN. In exposed node malware has installed successfully but not executed completely. Therefore, to stop the further transmission of malware in WSN apply a corrective measure on exposed state of nodes in time. Otherwise, these nodes become infectious with rate μ . Some of the exposed class of sensor nodes become dead with rate δ_2 because of malware attack.

• Sensor node of WSN gets compromise with malware and begin to malicious activity in the network. Infected nodes start to spread malware with surrounding nodes and increase their own energy dissipation. At the same time, malicious programs may decide not to continue attacking the infected node, and these infected nodes are only infectious and not destructive. The corrective measure applies in time and install anti-malware successfully manner, the node will be safely converted to the recovered state at the rate η . As the degree of damage from malicious programs increases, the node will die faster. Some of the infectious sensor nodes become dead with rate δ_3 because of attack of worms.

• Recovered state of nodes in WSN not only immune to malware attack but they have also level of high-energy. Sensor nodes of the network turn into recovered state infectious state. Some of the recovered class of sensor nodes become dead with rate δ_5 .

• Susceptible sensor nodes are adding in the in the system is at the rate of b.

Total number of sensor nodes in WSN at any time t is divided into five states. The N(t) number of nodes in the system these names are as Susceptible State S(t); Vaccinated V (t); Exposed State E(t)); Infectious State I(t); Recovered State R(t).

Therefore, mathematically satisfy the following equation N(t) = S(t) + E(t) + I(t) + V(t) + R(t), for any time $t \ge 0$.

TABLE 1. Description of used parameter.

S. No	Used Parameters	Description
1	1 41 41100015	
1	D	Susceptible recruitment rate
2	Υ	Vaccination rate
3	ρ	Rate of effectiveness of vaccination
4	μ	Rate of exposed node become
		infectious
5	η	Recovery rate of infectious node
6	δ_1	Mortality rate of susceptible nodes
7	δ_2	Mortality rate of exposed nodes
8	δ_3	Mortality rate of infectious nodes
9	δ_4	Mortality rate of vaccinated nodes
10	δ_5	Mortality rate of recovered nodes

To analyses the transmission dynamics of malware and state of change of sensor node presented by the set of differential equation. The state transition equations are written as:

$$\frac{dS}{dt} = b - \beta SI - (\delta_1 + \gamma) S$$

$$\frac{dE}{dt} = \beta SI + \rho \beta VI - (\delta_2 + \mu) E$$

$$\frac{dI}{dt} = \mu E - (\delta_3 + \eta) I$$

$$\frac{dV}{dt} = \gamma S - \rho \beta VI - \delta_4 V$$

$$\frac{dR}{dt} = \eta I - \delta_5 R \qquad (3.1)$$

To model a dimensionally consistent model, we assume that the initial value of the five compartments satisfy the initial conditions (ICs):

$$S(0) = S_0, \quad E(0) = E_0, \quad I(0) = I_0,$$

$$V(0) = V_0, \quad R(0) = R_0, \quad (3.2)$$

where S_0 , E_0 , I_0 , V_0 and R_0 are nonnegative real numbers.

In corporate of memory effect, we wish to convert the above model (3.1) into fractional order problem. To achieve this, equivalent integral form of (3.1) are given by

$$S(t) = S_0 + \int_0^t [b - \beta S(p)I(p) - (\delta_1 + \gamma) S(p)] dp$$

$$E(t) = E_0 + \int_0^t [\beta S(p)I(p) + \rho \beta V(p)I(p) - (\delta_2 + \mu) E(p)] dp$$

$$I(t) = I_0 + \int_0^t [\mu E(p) - (\delta_3 + \eta) I(p)] dp$$

$$V(t) = V_0 + \int_0^t [\gamma S(p) - \rho \beta V(p)I(p) - \delta_4 V(p)] dp$$

$$R(t) = R_0 + \int_0^t [\eta I(p) - \delta_5 R(p)] dp$$
(3.3a)

In order to include the influence of memory effect [66], we re-write (3.3a) in terms of time dependent integrals:

$$S(t) = S_0 + \int_{0}^{t} \varphi(t, p) \left[b - \beta S(p) I(p) - (\delta_1 + \gamma) S(p) \right] dp$$

$$E(t) = E_0 + \int_{0}^{t} \varphi(t, p) \left[\beta S(p) I(p) + \rho \beta V(p) I(p) - (\delta_2 + \mu) E(p) \right] dp$$

$$I(t) = I_0 + \int_{0}^{t} \varphi(t, p) \left[\mu E(p) - (\delta_3 + \eta) I(p) \right] dp$$

$$V(t) = V_0 + \int_{0}^{t} \varphi(t, p) \left[\gamma S(p) - \rho \beta V(p) I(p) - \delta_4 V(p) \right] dp$$

$$R(t) = R_0 + \int_{0}^{t} \varphi(t, p) \left[\eta I(p) - \delta_5 R(p) \right] dp$$
(3.3b)

where $\varphi(t, p)$ plays the role of a time – dependent memory kernel as a delta function $\delta(t, p)$ in the classical Markov process. The proper choice of $\varphi(t, p)$ can be a power – law correlation function, to incorporate long – term memory effect, which contributes to the evolution of the system. Thus, we can assume the non-local and nonsingular kernel as:

$$\varphi(t,p) = \frac{B(\alpha)}{1-\alpha} \mathbf{E}_{\alpha} \left(-\frac{\alpha(t-x)^{\alpha-1}}{1-\alpha}\right), \quad \alpha \in (0,1].$$

Putting this choice of $\varphi(t, p)$ into equation (3.3b) and using definition 2.1, we get:

$$S(t) - S_{0} = {}^{ABC}_{0}D_{t}^{-\alpha}[b - \beta SI - (\delta_{1} + \gamma)S]$$

$$E(t) - E_{0} = {}^{ABC}_{0}D_{t}^{-\alpha}[\beta SI + \rho\beta VI - (\delta_{2} + \mu)E]$$

$$I(t) - I_{0} = {}^{ABC}_{0}D_{t}^{-\alpha}[\mu E - (\delta_{3} + \eta)I]$$

$$V(t) - V_{0} = {}^{ABC}_{0}D_{t}^{-\alpha}[\gamma S - \rho\beta VI - \delta_{4}V]$$

$$R(t) - R_{0} = {}^{ABC}_{0}D_{t}^{-\alpha}[\eta I - \delta_{5}R]$$
(3.4)

Now, applying the Atangana–Baleanu fractional derivative of order α in Caputo sense on both sides of each equation in (3.4) and Lemma1, our fractional order epidemic model is given by

Both sides of the equations in (3.5) are assuming to be same in dimensions. Now, from here our full attention will be on this model (3.5) only. Fractional derivatives are very effective to describe memory effect in the modelling of dynamics of malware propagation in WSNs. The decay rate of the memory kernel $\varphi(t, p)$ depends on the value of α . A smaller value of α corresponds to more slowly decaying time – correlation functions. The fractional order α can be denoted as an index of memory of the given system, for detail concern [73].

V. EXISTENCE OF NON-NEGATIVE SOLUTION AND ITS UNIQUENESS

To discuss the existence and uniqueness of our model (3.5), we need to proof the following Theorem:

Theorem 4: The model (3.5) with the initial conditions (ICs) (3.2), has a unique positive solution for every $(S_0, E_0, I_0, V_0, R_0) \in \mathbb{R}^5_+$. Moreover, the compact set $Y = \{(S, E, I, V, R) \in \mathbb{R}^5_+ : 0 \le S + E + I + V + R \le \frac{b}{\delta}\}...$ (4.1) is a positively invariant set and having all solutions of the model (3.5) initiating in \mathbb{R}^5_+ . Where $\delta = Max\{\delta_1, \delta_2, \delta_3, \delta_4, \delta_5\}$

Proof:

Let

$$X(t) = \begin{bmatrix} S(t) \\ E(t) \\ I(t) \\ V(t) \\ R(t) \end{bmatrix}; X(0) = X_0 = \begin{bmatrix} S(0) \\ E(0) \\ I(0) \\ V(0) \\ R(0) \end{bmatrix} = \begin{bmatrix} S_0 \\ I_0 \\ V_0 \\ R_0 \end{bmatrix} W (X(t))$$
$$= \begin{bmatrix} W_1 (X(t)) \\ W_2 (X(t)) \\ W_3 (X(t)) \\ W_4 (X(t)) \\ W_5 (X(t)) \end{bmatrix} = \begin{bmatrix} b - \beta SI - (\delta_1 + \gamma) S \\ \beta SI + \rho \beta VI - (\delta_2 + \mu) E \\ \mu E - (\delta_3 + \eta) I \\ \gamma S - \rho \beta VI - \delta_4 V \\ \eta I - \delta_5 R \end{bmatrix}$$

Therefore, the model (3.5) with the ICs (3.2), can be written in matrix form as:

$${}^{BC}_{0}D^{\alpha}_{t}X(t) = W(X)$$
 with $X(0) = X_{0}$

The Jacobian matrix $J = \frac{\partial W}{\partial X} = \frac{\partial (W_1, W_2, W_3, W_4, W_5)}{\partial (S, E, I, V, R)}$ of W is continuous on \mathbb{R}^5_+ by [70], Remark 1.2.1] W is locally Lipschitz on \mathbb{R}^5_+ by [71], Remark 3.8], the model (3.5) with I.C. (3.2) has a unique solution for every $(S_0, E_0, I_0, V_0, R_0) \in \mathbb{R}^5_+$.

Now, We have to show that, for every $(S_0, E_0, I_0, V_0, R_0) \in \mathbb{R}^5_+$, the unique solution $((S, E, I, V, R) \in \mathbb{R}^5_+$ of model (3.5) with the ICs (3.2), is nonnegative. We deny this and discuss distinguish thirty cases.

Case 1: There exist t_S , t_E , t_I , t_V , $t_R \in (0, \infty)$ such that

1.1. $S(t_S) < 0$ and $S(t) \ge 0$ for all $t \in [0, t_S)$ (4.2)

- **1.2**. $E(t_E) < 0$ and $E(t) \ge 0$ for all $t \in [0, t_E)$ (4.3)
- **1.3**. $I(t_I) < 0$ and $I(t) \ge 0$ for all $t \in [0, t_I)$ (4.4)
- **1.4.** $V(t_V) < 0$ and $V(t) \ge 0$ for all $t \in [0, t_V)$ (4.5)
- **1.5.** $R(t_R) < 0$ and $R(t) \ge 0$ for all $t \in [0, t_R)$ (4.6)

Case 2: S(t) is non-negative on \mathbb{R}_+ and 4.3, 4.4, 4.5 and 4.6 holds (and other similar possible cases too)

Case 3: S(t), V(t) are non-negative on \mathbb{R}_+ and 4.3, 4.4, and 4.6 holds (and other similar possible cases too)

Case 4: S(t), E(t), I(t) are non-negative on \mathbb{R}_+ 4.5 and 4.6 holds (and other similar possible cases too)

Case 5: S(t), E(t), I(t), V(t) are non-negative on \mathbb{R}_+ and and 4.6 holds (and other similar possible cases too)

Using contradiction approach, we will prove that none of the above all possible combination of cases can occur.

If case 1 hold, then there are five possibilities arises, which is discussed below:

Subcase 1.1.For $t_S = \min\{t_S, t_E, t_I, t_V, t_R\}$, from the first equation in (3.5), we have ${}^{ABC}_{0}D^{\alpha}_{t}S(t) \ge -\beta SI (\delta_1 + \gamma) S \ge -k_1 S$ where, $k_1 = \beta^{\alpha} \max_{t \in [0, t_s]} I(t) + \delta_1 + \gamma > 0$; thus, after solving, ${}^{ABC}_{0} D^{\alpha}_t S(t) \ge -k_1 S$ we get, $S(t) \ge S(0)E_{\alpha}(-k_1t^{\alpha})$ for all $t \in [0, t_s]$. Hence, $S(t_s) \ge 0$. Which is against our assumption, $S(t_s) < 0$.

Subcase 1.2. For $t_E = \min\{t_S, t_E, t_I, t_V, t_R\}$ from the second equation in (3.5), we have ${}^{ABC}_{0}D^{\alpha}_{t}E(t) \ge -(\delta_{2} + \mu)E \ge$ k_2E where, $k_2 = (\delta_2 + \mu) > 0$; thus, after solving, ${}^{ABC}_{\Omega}D^{\alpha}_{t}E(t) \ge -k_{2}E \text{ we get, } E(t) \ge E(0)E_{\alpha}(-k_{2}t^{\alpha}) \text{ for all } t \in$ $[0, t_E]$. Hence, $E(t_E) \ge 0$. Which is against our assumption, $E(t_E) < 0.$

Subcase 1.3. For $t_I = \min\{t_S, t_E, t_I, t_V, t_R\}$, from the third equation in (3.5), we have ${}^{ABC}_{0}D^{\alpha}_{t}I(t) \ge -(\delta_{3}+\eta)I \ge$ k_3I where, $k_3 = (\delta_3 + \eta) > 0$; thus, after solving ${}^{ABC}_{0}D^{\alpha}_{t}I(t) \ge -k_{3}I$, we get, $I(t) \ge I(0)E_{\alpha}(-k_{3}t^{\alpha})$ for all $t \in$ $[0, t_I]$. Hence, $I(t_I) \ge 0$. This is again against our assumption, $I(t_I) < 0.$

Subcase 1.4. For $t_V = \min\{t_S, t_E, t_I, t_V, t_R\}$, from the fourth equation in (3.5), we have ${}^{ABC}_{0}D^{\alpha}_{t}V(t) \ge (\rho\beta I + \delta_4)V \ge -k_4V$ where, $k_4 = (\rho\beta max_{t\in[0,t_V]}I(t) +$ δ_4 > 0; thus, after solving ${}^{ABC}_{0}D^{\alpha}_t V(t) \ge -k_4 V$, we get, $V(t) \ge V(0)E_{\alpha}(-k_4t^{\alpha})$ for all $t \in [0, t_V]$. Hence, $V(t_V) \ge 0$. This is also against our assumption, $V(t_4) < 0$.

Subcase 1.5. For $t_R = \min\{t_S, t_E, t_I, t_V, t_R\}$, from the fifth equation in (3.5), we have ${}^{ABC}_{0}D^{\alpha}_{t}R(t) \ge \eta I - \delta_5 R \ge -k_5 R$ where, $k_5 = (\eta \max_{t \in [0, t_V]} I(t) + \delta_5) > 0$; thus, after solving $^{ABC}_{0}D_{t}^{\alpha}R(t) \geq -k_{5}R$, we get, $R(t) \geq R(0)E_{\alpha}(-k_{54} \text{ and } 5(4.5)(t^{\alpha}))$ for all $t \in [0, t_R]$. Hence, $R(t_R) \ge 0$. This is also against our assumption, $R(t_R) < 0$.

Hence, combining all the subcases (1.1) - (1.5) we reach the conclusion that case (1) does not possible. Now assuming case (2) holds. Proceeding in the similar way as in case we can also shows that case (2) cannot occur. Similar arguments can be used to show that none of the cases occur i.e. case (2) to case (5) and other all possible combinations of cases cannot occur. On the basis of the above analysis, we reached on the conclusion that the unique solution (S, E, I, V, R) is non-negative. This conclusion establishes the non-negativity of our solution. Now, we have to prove that set Y defined by (4.1) is positively invariant. Adding the equations in (3.5), we get:

Let $\delta = Max \{\delta_1, \delta_2, \delta_3, \delta_4, \delta_5\}$ and also assume that $\delta =$ $\delta_1 = \delta_2 = \delta_3 = \delta_4 = \delta_5$ for the sake of our convenience i.e. $D_t^{\alpha}M = b - \delta M$ (where, M = S + E + I + R + V solving above equation [24], remark 7.1]), we get:

$$M(t) = M(0)E_{\alpha} \left(-\delta t^{\alpha}\right) + b \int_{0}^{t} p^{\alpha-1}E'_{\alpha} \left(-\delta p^{\alpha}\right)dp$$
$$= \left(-\frac{b}{\delta} + M(0)\right)E_{\alpha} \left(-\delta t^{\alpha}\right) + \frac{b}{\delta}$$

Hence, we note that $E_{\alpha}(-\delta t^{\alpha}) \ge 0$. If $M(0) \le \frac{b}{\delta}$, we have $S+E+I+V+R = M(t) \le \frac{b}{\delta}$, Hence, Y is positively invariant. In the end, we successfully reached the conclusion that Y contains all solutions of the model (3.5) initially in \mathbb{R}^5_+ as $\lim_{t \to \infty} E_{\alpha} \left(-\delta^{\alpha} t^{\alpha} \right) = 0 \text{ (by Lemma 2.) we get } \lim_{t \to \infty} M(t) = \frac{b}{\delta}$ This completes the proof.

VI. EQUILIBRIUM AND STABILITY ANALYSIS **OF THE MODEL**

A. STABILITY ANALYSIS OF MALWARE - FREE **EQUILIBRIUM**

It is obvious that model (3.5) has a unique malware free equilibrium $W_0 = \left(\frac{b}{\delta_1 + \gamma}, 0, 0, \frac{\gamma b}{\delta_4(\delta_1 + \gamma)}, 0\right)$. Jacobian associated with the model (3.5) is:

$$= \begin{pmatrix} -I\beta - \gamma - \delta_1 & 0 & -\beta S & 0 & 0 \\ \beta I & -\mu - \delta_2 & \beta S + \beta \rho V & \beta I \rho & 0 \\ 0 & \mu & -\eta - \delta_3 & 0 & 0 \\ \gamma & 0 & -V\beta \rho & -\beta I \rho - \delta_4 & 0 \\ 0 & 0 & \eta & 0 & -\delta_5 \end{pmatrix}$$

At W_0 , the Jacobian

$$J(W_0) = \begin{pmatrix} -\gamma - \delta_1 & 0 & \frac{-\beta b}{\delta_1 + \gamma} & 0 & 0\\ 0 & -\mu - \delta_2 & \frac{\beta b}{\delta_1 + \gamma} + \frac{\gamma b \beta \rho}{\delta_4(\delta_1 + \gamma)} & 0 & 0\\ 0 & \mu & -\eta - \delta_3 & 0 & 0\\ \gamma & 0 & \frac{-\beta \rho \gamma b}{\delta_4(\delta_1 + \gamma)} & -\delta_4 & 0\\ 0 & 0 & \eta & 0 & -\delta_5 \end{pmatrix}$$

Its eigen values are $\lambda_1 = -(\gamma + \delta_1)$, $\lambda_2 = -\delta_4$, $\lambda_3 = -\delta_5 ,$

$$\lambda_{4} = \frac{1}{2(\gamma + \delta_{1})\delta_{4}} (-\gamma \eta \delta_{4} - \gamma \mu \delta_{4} \\ - \eta \delta_{1} \delta_{4} - \mu \delta_{1} \delta_{4} - \gamma \delta_{2} \delta_{4} \\ - \delta_{1} \delta_{2} \delta_{4} \\ - \gamma \delta_{3} \delta_{4} - \delta_{1} \delta_{3} \delta_{4} \\ - \sqrt{\gamma + \delta_{1}} \sqrt{\delta_{4}} \sqrt{(4b\beta\gamma\mu\rho + \gamma\eta^{2}\delta_{4} + 4b\beta\mu\delta_{4} \\ - 2\gamma\eta\mu\delta_{4} + \gamma\mu^{2}\delta_{4} + \eta^{2}\delta_{1}\delta_{4} \\ - 2\eta\mu\delta_{1}\delta_{4} + \mu^{2}\delta_{1}\delta_{4} - 2\gamma\eta\delta_{2}\delta_{4} + 2\gamma\mu\delta_{2}\delta_{4} \\ - 2\eta\delta_{1}\delta_{2}\delta_{4} + 2\mu\delta_{1}\delta_{2}\delta_{4} + \gamma\delta_{2}^{2}\delta_{4}$$

\

$$\begin{aligned} &+ \delta_{1}\delta_{2}^{2}\delta_{4} + 2\gamma\eta\delta_{3}\delta_{4} - 2\gamma\mu\delta_{3}\delta_{4} \\ &+ 2\eta\delta_{1}\delta_{3}\delta_{4} - 2\mu\delta_{1}\delta_{3}\delta_{4} - 2\gamma\delta_{2}\delta_{3}\delta_{4} - 2\delta_{1}\delta_{2}\delta_{3}\delta_{4} \\ &+ \gamma\delta_{3}^{2}\delta_{4} + \delta_{1}\delta_{3}^{2}\delta_{4})) \\ \lambda_{5} = \frac{1}{2(\gamma + \delta_{1})\delta_{4}}(-\gamma\eta\delta_{4} - \gamma\mu\delta_{4} \\ &- \eta\delta_{1}\delta_{4} - \mu\delta_{1}\delta_{4} - \gamma\delta_{2}\delta_{4} - \delta_{1}\delta_{2}\delta_{4} \\ &- \gamma\delta_{3}\delta_{4} - \delta_{1}\delta_{3}\delta_{4} + \sqrt{\gamma + \delta_{1}}\sqrt{\delta_{4}}\sqrt{(4b\beta\gamma\mu\rho + \gamma\eta^{2}\delta_{4} \\ &+ 4b\beta\mu\delta_{4} - 2\gamma\eta\mu\delta_{4} + \gamma\mu^{2}\delta_{4} + \eta^{2}\delta_{1}\delta_{4} - 2\eta\mu\delta_{1}\delta_{4} \\ &+ \mu^{2}\delta_{1}\delta_{4} - 2\gamma\eta\delta_{2}\delta_{4} + 2\gamma\mu\delta_{2}\delta_{4} - 2\eta\delta_{1}\delta_{2}\delta_{4} \\ &+ 2\mu\delta_{1}\delta_{2}\delta_{4} + \gamma\delta_{2}^{2}\delta_{4} \\ &+ 2\eta\delta_{1}\delta_{3}\delta_{4} - 2\mu\delta_{1}\delta_{3}\delta_{4} \\ &- 2\gamma\delta_{2}\delta_{3}\delta_{4} - 2\delta_{1}\delta_{2}\delta_{3}\delta_{4} + \gamma\delta_{3}^{2}\delta_{4} + \delta_{1}\delta_{3}^{2}\delta_{4})) \end{aligned}$$

The first four eigen values are obviously negative and last one will be negative if and only if $(-\gamma\eta\delta_4 - \gamma\mu\delta_4 - \eta\delta_1\delta_4 - \mu\delta_1\delta_4 - \gamma\delta_2\delta_4 - \delta_1\delta_2\delta_4 - \gamma\delta_3\delta_4 - \delta_1\delta_3\delta_4 + \sqrt{\gamma + \delta_1}\sqrt{\delta_4}\sqrt{(4b\beta\gamma\mu\rho + \gamma\eta^2\delta_4 + 4b\beta\mu\delta_4 - 2\gamma\eta\mu\delta_4 + \gamma\mu^2\delta_4 + \eta^2\delta_1\delta_4 - 2\eta\mu\delta_1\delta_4 + \mu^2\delta_1\delta_4 - 2\gamma\eta\delta_2\delta_4 + 2\gamma\mu\delta_2\delta_4 - 2\eta\delta_1\delta_2\delta_4 + 2\mu\delta_1\delta_2\delta_4 + \gamma\delta_2^2\delta_4 + \delta_1\delta_2^2\delta_4 + 2\gamma\delta_3\delta_4 - 2\gamma\delta_3\delta_4 - 2\gamma\delta_3\delta_4 - 2\gamma\delta_3\delta_4 - 2\gamma\delta_3\delta_4 + \gamma\delta_3^2\delta_4 + \delta_1\delta_3^2\delta_4)) < 0$ or $R_0^{th} = \frac{\delta_4b\beta\mu + b\beta\gamma\rho\mu}{(\delta_1 + \gamma)(\mu + \delta_2)(\eta + \delta_3)\delta_4} < 1$, where R_0^{th} is a dimensionless basic reproduction number.

Using Lemma 3 and above analysis, the following result is immediate:

Theorem 5.1: The malware free equilibrium W_0 is locally asymptotically stable if $R_0^{th} < 1$ and unstable if $R_0^{th} > 1$.

From Above said theorem it is clear that if the initial values $(S_0, E_0, I_0, V_0, R_0) \in \mathbb{R}^5_+$ are sufficiently close to W_0 , the unique solution of the model (3.5) with Initial Conditions (3.2), i.e. converges to W_0 . But, here we cannot depict the region of attraction, for that we discuss the next result:

Theorem 5.2: Consider $R_0^{th} < 1$. The compact set $Y = \{(S, E, I, V, R) \in \mathbb{R}^5_+ : 0 \le S + E + I + V + R \le \frac{b}{\delta}\},\$ is the region of attraction of the user-free equilibrium W_0 .

Proof: We have to show that, for every $((S_0, E_0, I_0, V_0, R_0) \in Y$, the unique solution (S, E, I, V, R) of model (3.5) with the Initial Conditions (ICs) (3.2) converges to W_0 . Let consider the Lyapunovfunction

$$L(t) = \frac{1}{2}I^{2} + \frac{1}{2}\left(s - \frac{b}{\delta} + E + I + V + R\right)^{2}$$
(5.3)

By theorem 5.1, Υ is nonnegative invariant. Since, $(S_0, E_0, I_0, V, R_0) \in \Upsilon$, we have $s \leq \frac{b}{\delta_1 + \gamma}$

Therefore, Lemma 4. and model (3.5), we get:

$$ABC_{0}D_{t}^{\alpha}L \leq I(ABC_{0}D_{t}^{\alpha}I(t)) + \left(S - \frac{b}{\delta} + E + I + V + R\right) \times ABC_{0}D_{t}^{\alpha}\left(S - \frac{b}{\delta} + E + I + V + R\right) = I(\mu E - (\delta_{3} + \eta)I)$$

$$+ \left(S - \frac{b}{\delta} + E + I + V + R\right)$$

× $(b - \delta_1 S - \delta_2 E - \delta_3 I - \delta_4 V - \delta_5 R)$
 $\leq I (\mu E - (\delta_3 + \eta) I)$
+ $\left(S - \frac{b}{\delta} + I + V + R\right) (b - \delta(S + E + I + V + R))$

 $(as \ \delta = max \{\delta_1, \delta_2, \delta_3, \delta_4, \delta_5\})$

$$\leq I\left(\mu E - (\delta_3 + \eta)I\right) - \delta\left(S - \frac{b}{\delta} + E + I + V + R\right)^2$$

 $(\therefore S \leq \frac{b}{\delta_1 + \gamma})$

$$ABC_{0}D_{t}^{\alpha}L \leq 0 \text{ if and only if } I(\mu E - (\delta_{3} + \eta)I)$$
$$- \delta \left(S - \frac{b}{\delta} + E + I + V + R\right)^{2} \leq 0$$

or $I(\mu E - (\delta_3 + \eta)I) \leq \delta \left(S - \frac{b}{\delta} + E + I + V + R\right)^2$ Moreover, ${}^{ABC}_{0}D^{\alpha}_{t}L \leq 0$ if and only if $S + E + I + V + R = \frac{b}{\delta}$ and I = 0.

i.e. $W = \{(S, E, I, V, R); {}^{ABC}_{0}D^{\alpha}_{t}L = 0\} = \{(S, E, I, V, R); S + E + I + V + R = \frac{b}{\delta}\}$

By Lemma 5., every solution of model (3.5) initiating in Y tend to the largest invariant set in W. Thus, $\lim_{t\to\infty} I(t) = 0$. for I = 0, from (3.5) we get

$${}^{ABC}_{0}D^{\alpha}_{t}S(t) = b - (\delta_{1} + \gamma)S$$

$${}^{ABC}_{0}D^{\alpha}_{t}E(t) = - (\delta_{2} + \mu)E$$

$${}^{ABC}_{0}D^{\alpha}_{t}V(t) = \gamma S - \delta_{4}V$$

$${}^{ABC}_{0}D^{\alpha}_{t}R(t) = -\delta_{5}R \qquad (5.4)$$

In similar manner to obtaining (3.5), we can find the solution of (5.4), also thus, after solving, we get, (similar to (3.5))

$$S(t) = \left(-\frac{b}{(\delta_1 + \gamma)} + S(0)\right) E_{\alpha} \left(-((\delta_1 + \gamma))t^{\alpha}\right) + \frac{b}{(\delta_1 + \gamma)} E(t) = E(0)E_{\alpha} \left(-(\delta_2 + \mu)t^{\alpha}\right) V(t) = \left(-\frac{\gamma S(0)}{\delta_4} + V(0)\right) E_{\alpha} \left(-\delta_4 t^{\alpha}\right) + \frac{\gamma}{\delta_4} \frac{b}{(\delta_1 + \gamma)} R(t) = r(0)E_{\alpha} \left(-\delta_5 t^{\alpha}\right)$$
(5.5)

By Lemma 2.

$$\lim_{t \to \infty} \mathbf{S}(t) = \frac{b}{(\delta_1 + \gamma)}, \lim_{t \to \infty} \mathbf{E}(t) = 0, \quad \lim_{t \to \infty} \mathbf{V}(t)$$
$$= \frac{\gamma}{\delta_4} \frac{b}{(\delta_1 + \gamma)} \text{ and } \lim_{t \to \infty} \mathbf{R}(t) = 0$$

Thus, $(S, E, I, V, R) \rightarrow W_0$ as $t \rightarrow \infty$. This Thus, $(S, E, I, V, R) \rightarrow W_0$ as $t \rightarrow \infty$.

This completes the proof of the theorem.

B. STABILITY ANALYSIS OF ENDEMIC EQUILIBRIUM

If $R_0^{th} > 1$, Firstly, we show the existence of a unique endemic equilibrium of the model (3.5).

Theorem 5.3: If $R_0^{th} > 1$. The model (3.5) has a unique endemic equilibrium $W^* = (S^*, E^*, I^*, V^*, R^*)$, where $S^* = \frac{b}{\beta I^* + (\delta_1 + \gamma)}, E^* = \frac{R^* \delta_5(\delta_3 + \eta)}{\mu \eta}, I^* = \frac{\delta_5 R^*}{\eta}, V^* = \frac{\gamma S^*}{\rho \beta I^* + \delta_4}$ and R^* is the unique positive solution of the cubic equation in R.

Proof.

From Definition 2.3, an endemic equilibrium point of the model (3.5) satisfies the equations

$$0 = b - \beta S^* I^* - (\delta_1 + \gamma) S^*$$

$$0 = \beta S^* I^* + \rho \beta V^* I^* - (\delta_2 + \mu) E^*$$

$$0 = \mu E^* - (\delta_3 + \eta) I^*$$

$$0 = \gamma S^* - \rho \beta V^* I^* - \delta_4 V^*$$

$$0 = \eta I^* - \delta_5 R^*$$
(5.6)

After straight forward calculation, endemic equilibrium points are given by

$$S^{*} = \frac{\eta b}{(\beta \delta_{5} R^{*} + (\delta_{1} + \gamma) \eta)}, E^{*} = \frac{(\delta_{3} + \eta) \delta_{5} R^{*}}{\eta \mu},$$
$$I^{*} = \frac{\delta_{5} R^{*}}{\eta}, \quad V^{*} = \frac{\gamma S^{*}}{(\rho \beta \delta_{5} R^{*} + \delta_{4} \eta)},$$

where R^* is given by the equation

$$\frac{\beta b\eta}{\beta \delta_5 R^* + \eta (\delta_1 + \gamma)} \frac{\delta_5 R^*}{\eta} - \frac{R^* \delta_5 (\delta_3 + \eta)}{\mu \eta} - \frac{\delta_3 \delta_5 R^*}{\eta} + \frac{\gamma b\eta}{\beta \delta_5 R^* + \eta (\delta_1 + \gamma)} - \frac{\delta_4 \gamma b (\eta)^2}{[\rho \beta R^* \delta_5 + \delta_4 \eta] [\beta \delta_5 R^* + \eta (\delta_1 + \gamma)]} - \delta_5 R^* = 0$$
(5.7)

From equation (5.7) it is clear that R^* have non-negative value if $R_0^{th} = \frac{\delta_4 b\beta\mu + b\beta\gamma\rho\mu}{(\delta_1 + \gamma)(\mu + \delta_2)(\eta + \delta_3)\delta_4} > 1$. which proves the theorem (5.3).

Theorem 5.4: The endemic equilibrium $W^* = (S^*, E^*, I^*, V^*, R^*)$ is globally asymptotically stable.

Proof: Let $\psi(x) = x - x^* - x^* \ln \frac{x}{x^*}$ Define the LyapunovfunctionL ₁(t) by

$$L_{1}(t) = a_{1}\psi_{1} (S(t)) + a_{2}\psi_{2} (E(t)) + a_{3}\psi_{3} (I(t)) + a_{4}\psi_{4} (V(t)) + a_{5}\psi_{5}(R(t)), where $a_{j} > 0, \ j = 1, 2, 3, 4, 5$
$$L_{1}(t) = a_{1} \left(S - S^{*} - s^{*} \ln \frac{S}{S^{*}} \right) + a_{2} \left(E - E^{*} - E^{*} \ln \frac{E}{E^{*}} \right) + a_{3}(I - I^{*} - I^{*} \ln \frac{I}{I^{*}}) + a_{4}(V - V^{*} - V^{*} \ln \frac{V}{V^{*}}) + a_{5}(R - R^{*} - R^{*} \ln \frac{R}{R^{*}})$$
(5.8)$$

Now, we calculate the α – order Caputo derivatives of ψ_j , j = 1, 2, 3, 4, 5 by Lemma 4. From the first Equation of (5.6), we have

$$b = \beta S^* I^* + (\delta_1 + \gamma) S^*$$

Using this and the first equation of (3.5), we obtain

$$ABC_{0}D_{t}^{\alpha}\psi_{1}(S) \leq \frac{S-S^{*}}{S}ABC_{0}D_{t}^{\alpha}S(t)$$

$$= -\frac{(S-S^{*})^{2}}{S}\left(\beta I^{*} + (\delta_{1}+\gamma)\right)$$

$$+ \beta \left(S-S^{*}\right)\left(I^{*}-I\right)$$
(5.9)

From second equation of (5.6)

$$\beta S^* I^* + \rho \beta V^* I^* = (\delta_2 + \mu) E^*$$

Using this and the second equation in (3.5), we obtain

$${}^{ABC}_{0}D^{\alpha}_{t}\psi_{2}\left(E(t)\right) \leq \frac{(E-E^{*})}{E}{}^{ABC}_{0}D^{\alpha}_{t}E(t)$$

$$= \frac{(E-E^{*})}{E}\left[\beta\frac{(\mathrm{SI}E^{*}-I^{*}S^{*}E)}{E^{*}}\right]$$

$$+ \rho\beta\frac{(\mathrm{VI}E^{*}-I^{*}V^{*}E)}{E^{*}}] \quad (5.10)$$

Now, from third equation of (5.6), we get

$$0 = \mu E^* - (\delta_3 + \eta) I^*$$

Using this and from third equation of (3.5), we get

$${}^{ABC}_{0}D^{\alpha}_{t}\psi_{3}(I) \leq \frac{(I-I^{*})}{I}{}^{ABC}_{0}D^{\alpha}_{t}I(t)$$
$$= \frac{(I-I^{*})}{I} \left[\mu(E-E^{*}) - \frac{\mu E^{*}}{I^{*}}(I-I^{*})\right]$$
(5.11)

From fourth equation of (5.6), we get

$$\gamma S^* = \rho \beta V^* I^* + \delta_4 V^*$$

Using this and from fourth equation of (3.5), we get

$$\frac{AB_{0}^{C}D_{t}^{\alpha}\psi_{4}(V)}{\leq \frac{(V-V^{*})}{V}AB_{0}^{C}D_{t}^{\alpha}V(t)} = \frac{(V-V^{*})}{R}\left(S\frac{\rho\beta V^{*}I^{*}+\delta_{4}V^{*}}{S^{*}}-\rho\beta VI-\delta_{4}V\right) (5.12)$$

From fifth equation of (5.6), we get

$$\eta I^* = \delta_5 R^*$$

Using this and from fifth equation of (3.5), we get

$$ABC_{0}D_{t}^{\alpha}\psi_{5}(R) \leq \frac{(R-R^{*})}{R}ABC_{0}D_{t}^{\alpha}R(t)$$
$$= \frac{(R-R^{*})}{R}(\frac{\delta_{5}R^{*}I - \delta_{5}RI^{*}}{I^{*}}) \qquad (5.13)$$

From equation (5.3) to (5.13), it follows that ${}^{ABC}_{0}D^{\alpha}_{t}L_{1}(t) \leq 0$ and ${}^{ABC}_{0}D^{\alpha}_{t}L_{1}(t) = 0$ if and only if $S = S^{*}$, $E = E^{*}$, $I = I^{*}$, $V = V^{*}$ and $R = R^{*}$, Hence, the largest

invariant set $\{(S, E, I, V, R); {}^{ABC}_{0}D^{\alpha}_{t}L = 0\}$ is the singleton set W^* . From Theorem 4.1, we found that the set Y attracts all solutions of the model (3.5) initiating in \mathbb{R}^5_+ . By Lemma 5, W^* is global asymptotically stable. This proofs the theorem.

VII. NUMERICAL SOLUTION AND ITS ANALYSIS

Here, we find the numerical solutions of the considered model (3.1). Then the numerical results are gained through the proposed scheme. To achieve this purpose, we apply the fractional Adams Bashforth method [60] to approximate the AB fractional integral.

The equation (3.1) can be converted to a fractional integral equation by applying the fundamental theorem of fractional calculus [[24], [42], [47]] as follows:

$$\begin{aligned} S_n(t) &= S_n(0) + \frac{(1-\alpha)}{B(\alpha)} \{b - \beta S_n I_n - (\delta_1 + \gamma) S_n\} \\ &+ \frac{\alpha}{B(\alpha)} \prod_{i=0}^{t} (t-i)^{\alpha-1} \\ &\times \{b - \beta S_n I_n - (\delta_1 + \gamma) S_n\} dl \\ E_n(t) &= E_n(0) + \frac{(1-\alpha)}{B(\alpha)} \{\beta S_n I_n + \rho \beta V_n I_n - \mu E_n - \delta E_n\} \\ &+ \frac{\alpha}{B(\alpha)} \prod_{i=0}^{t} (t-i)^{\alpha-1} \\ &\times \{\beta S_n I_n + \rho \beta V_n I_n - \mu E_n - \delta_2 E_n\} dl \\ I_n(t) &= I_n(0) + \frac{(1-\alpha)}{B(\alpha)} \{\mu E_n - (\delta_3 + \eta) I_n\} \\ &+ \frac{\alpha}{B(\alpha)} \prod_{i=0}^{t} (t-i)^{\alpha-1} \{\mu E_n - (\delta_3 + \eta) I_n\} dl \\ V_n(t) &= V_n(0) + \frac{(1-\alpha)}{B(\alpha)} \{\gamma S_n - \rho \beta V_n I_n - \delta_4 V_n\} \\ &+ \frac{\alpha}{B(\alpha)} \prod_{i=0}^{t} (t-i)^{\alpha-1} \\ &\times \{\gamma S_n - \rho \beta V_n I_n - \delta_4 V_n\} dl \\ R_n(t) &= R_n(0) + \frac{(1-\alpha)}{B(\alpha)} \{\eta I_n - \delta_5 R_n\} \\ &+ \frac{\alpha}{B(\alpha)} \prod_{i=0}^{t} (t-i)^{\alpha-1} \{\eta I_n - \delta_5 R_n\} dl \end{aligned}$$
(6.1)

For the sake of convenience, let us assume

$$H_{n,1}(t) = b - \beta S_n(t)I_n(t) - (\delta_1 + \gamma) S_n(t),$$

$$H_{n,2}(t) = \beta S_n(t)I_n(t) + \rho\beta V_n(t)I_n(t) - \mu E_n(t) - \delta E_n(t),$$

$$H_{n,3}(t) = \mu E_n(t) - (\delta_3 + \eta) I_n(t),$$

$$H_{n,4}(t) = \gamma S_n - \rho\beta V_n(t)I_n(t) - \delta_4 V_n(t),$$

$$H_{n,5}(t) = \eta I_n(t) - \delta_5 R_n(t)$$
(6.2)

The expressions $H_{n,1}(t)$, $H_{n,2}(t)$, $H_{n,3}(t)$, $H_{n,4}(t)$, and $H_{n,5}(t)$ are said to satisfy the Lipschitz condition if and only if $S_n(t)$, $E_n(t)$, $I_n(t)$, $V_n(t)$ and $R_n(t)$ have an upper bound. Let $S_n(t)$ and $S_m(t)$ be two functions, then we get

$$\begin{aligned} \left\| H_{n,1}(t) - H_{m,1}(t) \right\| \\ &= \left\| -\beta(S_n - S_m) I_n - (\delta_1 + \gamma) (S_n - S_m) \right\| \\ &= \left\| -(\beta I_n + (\delta_1 + \gamma)) (S_n - S_m) \right\| \\ &\leq \left\| -(\beta I_n + (\delta_1 + \gamma)) \right\| \left\| (S_n - S_m) \right\| \\ &\leq \varepsilon_1 \left\| (S_n - S_m) \right\| \end{aligned}$$
(6.3)

Where $\varepsilon_1 = \left\| -(\beta max_{t \in [0, t_S]} I_n(t) + (\delta_1 + \gamma)) \right\|$. Hence, we have

$$\|H_{n,1}(t) - H_{m,1}(t)\| \le \varepsilon_1 \|(S_n - S_m)\|$$
 (6.4)

Similarly, we can obtain

$$\begin{aligned} \left\| H_{n,2}(t) - H_{m,2}(t) \right\| &\leq \varepsilon_2 \left\| (E_n - E_m) \right\| \\ \left\| H_{n,3}(t) - H_{m,3}(t) \right\| &\leq \varepsilon_3 \left\| (I_n - I_m) \right\| \\ \left\| H_{n,4}(t) - H_{m,4}(t) \right\| &\leq \varepsilon_4 \left\| (V_n - V_m) \right\| \\ \left\| H_{n,5}(t) - H_{m,5}(t) \right\| &\leq \varepsilon_5 \left\| (R_n - R_m) \right\| \end{aligned}$$
(6.5)

Thus, the Lipschitz condition is satisfied for all the five functions $H_{n,1}(t)$, $H_{n,2}(t)$, $H_{n,3}(t)$, $H_{n,4}(t)$, and $H_{n,5}(t)$ where ε_1 , ε_2 , ε_3 , ε_4 and ε_5 are the corresponding Lipschitz constant

Now, using equation (6.2) in equation (6.1) we get:

$$S_{n}(t) = S_{n}(0) + \frac{(1-\alpha)}{B(\alpha)} \{b - \beta S_{n}I_{n} - (\delta_{1} + \gamma) S_{n}\} + \frac{\alpha}{B(\alpha)} \prod_{0}^{t} (t-l)^{\alpha-1}H_{n,1}(l)dl E_{n}(t) = E_{n}(0) + \frac{(1-\alpha)}{B(\alpha)} \{\beta S_{n}I_{n} + \rho\beta V_{n}I_{n} - \mu E_{n} - \delta E_{n}\} \times \frac{\alpha}{B(\alpha)} \prod_{0}^{t} (t-l)^{\alpha-1}H_{n,2}(l)dl I_{n}(t) = I_{n}(0) + \frac{(1-\alpha)}{B(\alpha)} \{\mu E_{n} - (\delta_{3} + \eta) I_{n}\} + \frac{\alpha}{B(\alpha)} \prod_{0}^{t} (t-l)^{\alpha-1}H_{n,3}(l)dl V_{n}(t) = V_{n}(0) + \frac{(1-\alpha)}{B(\alpha)} \{\gamma S_{n} - \rho\beta V_{n}I_{n} - \delta_{4}V_{n}\} + \frac{\alpha}{B(\alpha)} \prod_{0}^{t} (t-l)^{\alpha-1}H_{n,4}(l)dl R_{n}(t) = R_{n}(0) + \frac{(1-\alpha)}{B(\alpha)} \{\eta I_{n} - \delta_{5}R_{n}\} + \frac{\alpha}{B(\alpha)} \prod_{0}^{t} (t-l)^{\alpha-1}H_{n,5}(l)dl$$
(6.6)

VOLUME 11, 2023

To start an iterative scheme, at given point $t = t_{k+1}$, for $k = 0, 1, 2, \ldots$ the above equation is reformulated as:

$$S_{n}(t_{k+1}) = S_{n}(0) + \frac{(1-\alpha)}{B(\alpha)}H_{n,1}(t_{k}) + \frac{\alpha}{B(\alpha)\Gamma(\alpha)}\sum_{j=0}^{j=k}\int_{t_{j}}^{t_{j+1}}(t_{k+1}-l)^{\alpha-1}H_{n,1}(l)dl E_{n}(t_{k+1}) = E_{n}(0) + \frac{(1-\alpha)}{B(\alpha)}H_{n,2}(t_{k}) + \frac{\alpha}{B(\alpha)\Gamma(\alpha)}\sum_{j=0}^{j=k}\int_{t_{j}}^{t_{j+1}}(t_{k+1}-l)^{\alpha-1}H_{n,2}(l)dl I_{n}(t_{k+1}) = I_{n}(0) + \frac{(1-\alpha)}{B(\alpha)}H_{n,3}(t_{k})$$

$$B(\alpha) + \frac{\alpha}{B(\alpha) \Gamma(\alpha)} \sum_{j=0}^{j=k} \int_{t_j}^{t_{j+1}} (t_{k+1} - l)^{\alpha - 1} H_{n,3}(l) dl$$

$$V_n(t_{k+1}) = V_n(0) + \frac{(1 - \alpha)}{B(\alpha)} H_{n,4}(t_k) + \frac{\alpha}{B(\alpha) \Gamma(\alpha)} \sum_{j=0}^{j=k} \int_{t_j}^{t_{j+1}} (t_{k+1} - l)^{\alpha - 1} H_{n,4}(l) dl$$

$$R_{n}(t_{k+1}) = R_{n}(0) + \frac{1}{B(\alpha)} H_{n,5}(t_{k}) + \frac{\alpha}{B(\alpha) \Gamma(\alpha)} \sum_{j=0}^{j=k} \int_{t_{j}}^{t_{j+1}} (t_{k+1} - l)^{\alpha - 1} H_{n,5}(l) dl$$
(6.7)

Within the interval $[t_j, t_{j+1}]$, we can utilize the two points interpolation for approximate the functions $H_{n,1}(l)$, $H_{n,2}(l)$, $H_{n,3}(l)$, $H_{n,4}(l)$ and $H_{n,5}(l)$ inside the above integral.

$$H_{n,1}(l) \cong \frac{H_{n,1}(t_j)}{h} (l - t_{j-1}) - \frac{H_{n,1}(t_{j-1})}{h} (l - t_j)$$

$$H_{n,2}(l) \cong \frac{H_{n,2}(t_j)}{h} (l - t_{j-1}) - \frac{H_{n,2}(t_{j-1})}{h} (l - t_j)$$

$$H_{n,3}(l) \cong \frac{H_{n,3}(t_j)}{h} (l - t_{j-1}) - \frac{H_{n,3}(t_{j-1})}{h} (l - t_j)$$

$$H_{n,4}(l) \cong \frac{H_{n,4}(t_j)}{h} (l - t_{j-1}) - \frac{H_{n,4}(t_{j-1})}{h} (l - t_j)$$

$$H_{n,5}(l) \cong \frac{H_{n,5}(t_j)}{h} (l - t_{j-1}) - \frac{H_{n,5}(t_{j-1})}{h} (l - t_j) (6.8)$$

Using equation (6.4) and (6.5) in (6.3) and on simplification we get:

$$S_{n}(t_{k+1}) = S_{n}(0) + \frac{(1-\alpha)}{B(\alpha)} H_{n,1}(t_{k}) + \frac{\alpha}{B(\alpha) \Gamma(\alpha)} \sum_{j=0}^{j=k} \left(\frac{H_{n,1}(t_{j})}{h} I_{j-1}^{\alpha} - \frac{H_{n,1}(t_{j-1})}{h} I_{j}^{\alpha} \right)$$

$$E_{n}(t_{k+1}) = E_{n}(0) + \frac{(1-\alpha)}{B(\alpha)} H_{n,2}(t_{k}) + \frac{\alpha}{B(\alpha) \Gamma(\alpha)} \sum_{j=0}^{j=k} \left(\frac{H_{n,2}(t_{j})}{h} I_{j-1}^{\alpha} - \frac{H_{n,2}(t_{j-1})}{h} I_{j}^{\alpha} \right)$$
(6.9)

$$I_{n}(t_{k+1}) = I_{n}(0) + \frac{(1-\alpha)}{B(\alpha)} H_{n,3}(t_{k}) + \frac{\alpha}{B(\alpha) \Gamma(\alpha)} \sum_{j=0}^{j=k} \left(\frac{H_{n,3}(t_{j})}{h} I_{j-1}^{\alpha} - \frac{H_{n,3}(t_{j-1})}{h} I_{j}^{\alpha} \right) V_{n}(t_{k+1})$$

$$= V_n(0) + \frac{(1-\alpha)}{B(\alpha)} H_{n,4}(t_k) + \frac{\alpha}{B(\alpha) \Gamma(\alpha)} \sum_{j=0}^{j=k} \left(\frac{H_{n,4}(t_j)}{h} I_{j-1}^{\alpha} - \frac{H_{n,4}(t_{j-1})}{h} I_j^{\alpha} \right) R_n(t_{k+1})$$

$$= R_n(0) + \frac{(1-\alpha)}{B(\alpha)} H_{n,5}(t_k) + \frac{\alpha}{B(\alpha) \Gamma(\alpha)} \sum_{j=0}^{j=k} \left(\frac{H_{n,5}(t_j)}{h} I_{j-1}^{\alpha} - \frac{H_{n,5}(t_{j-1})}{h} I_j^{\alpha} \right)$$

where,

$$I_{j-1}^{\alpha} = \int_{t_j}^{t_{j+1}} (l - t_{j-1}) (t_{k+1} - l)^{\alpha - 1} dl$$
$$I_j^{\alpha} = \int_{t_j}^{t_{j+1}} (l - t_j) (t_{k+1} - l)^{\alpha - 1} dl$$

On further simplification of the integrals I_{j-1}^{α} and I_j^{α} , and assuming, $t_j = jh$, we can easily find that

$$I_{j-1}^{\alpha} = \frac{h^{\alpha+1}}{\alpha (\alpha+1)} \left[(k+1-j)^{\alpha} (k-j+2+\alpha) - (k-j)^{\alpha} (k-j+2+2\alpha) \right]$$
(6.10)
$$I_{j}^{\alpha} = \frac{h^{\alpha+1}}{\alpha (\alpha+1)} \left[(k+1-j)^{\alpha+1} - (k-j)^{\alpha} (k-j+1+\alpha) \right]$$
(6.11)

Using values from equations (6.10) and (6.11) in equation (6.9) and after simplification, we get

$$S_{n}(t_{k+1}) = S_{n}(0) + \frac{(1-\alpha)}{B(\alpha)} H_{n,1}(t_{k}) + \frac{\alpha}{B(\alpha)} \sum_{j=0}^{j=k} \times \left(\frac{H_{n,1}(t_{j})}{\Gamma(\alpha+2)} h^{\alpha} \left[(k+1-j)^{\alpha} (k-j+2+\alpha) - (k-j)^{\alpha} (k-j+2+2\alpha) \right] - \frac{H_{n,1}(t_{j-1})}{\Gamma(\alpha+2)} h^{\alpha}$$

49053

$$\left\lfloor (k+1-j)^{\alpha+1} - (k-j)^{\alpha}(k-j+1+\alpha) \right\rfloor \right)$$
(6.12)

$$E_{n}(t_{k+1}) = E_{n}(0) + \frac{(1-\alpha)}{B(\alpha)} H_{n,2}(t_{k}) + \frac{\alpha}{B(\alpha)} \sum_{j=0}^{j=k} \\ \times \left(\frac{H_{n,2}(t_{j})}{\Gamma(\alpha+2)} h^{\alpha} \left[(k+1-j)^{\alpha} (k-j+2+\alpha) - (k-j)^{\alpha} (k-j+2+\alpha) \right] - \frac{H_{n,2}(t_{j-1})}{\Gamma(\alpha+2)} h^{\alpha} \\ \left[(k+1-j)^{\alpha+1} - (k-j)^{\alpha} (k-j+1+\alpha) \right] \right)$$
(6.13)

$$I_{n}(t_{k+1}) = I_{n}(0) + \frac{(1-\alpha)}{B(\alpha)} H_{n,3}(t_{k}) + \frac{\alpha}{B(\alpha)} \sum_{j=0}^{j=k} \\ \times \left(\frac{H_{n,3}(t_{j})}{\Gamma(\alpha+2)} h^{\alpha} \left[(k+1-j)^{\alpha} (k-j+2+\alpha) - (k-j)^{\alpha} (k-j+2+2\alpha) \right] - \frac{H_{n,3}(t_{j-1})}{\Gamma(\alpha+2)} h^{\alpha} \\ \left[(k+1-j)^{\alpha+1} - (k-j)^{\alpha} (k-j+1+\alpha) \right] \right)$$
(6.14)

 $V_n(t_{k+1})$

`

$$= V_{n}(0) + \frac{(1-\alpha)}{B(\alpha)} H_{n,4}(t_{k}) + \frac{\alpha}{B(\alpha)} \sum_{j=0}^{j=k} \left\{ \frac{H_{n,4}(t_{j})}{\Gamma(\alpha+2)} h^{\alpha} \left[(k+1-j)^{\alpha} (k-j+2+\alpha) - (k-j)^{\alpha} (k-j+2+2\alpha) \right] - \frac{H_{n,4}(t_{j-1})}{\Gamma(\alpha+2)} h^{\alpha} \left[(k+1-j)^{\alpha+1} - (k-j)^{\alpha} (k-j+1+\alpha) \right] \right\}$$
(6.15)

$$R_{n}(t_{k+1}) = R_{n}(0) + \frac{(1-\alpha)}{B(\alpha)} H_{n,5}(t_{k}) + \frac{\alpha}{B(\alpha)} \sum_{j=0}^{j=k} \left(\frac{H_{n,5}(t_{j})}{\Gamma(\alpha+2)} h^{\alpha} \left[(k+1-j)^{\alpha} (k-j+2+\alpha) - (k-j)^{\alpha} (k-j+2+2\alpha) \right] - \frac{H_{n,5}(t_{j-1})}{\Gamma(\alpha+2)} h^{\alpha} \times \left[(k+1-j)^{\alpha+1} - (k-j)^{\alpha} (k-j+1+\alpha) \right] \right) (6.16)$$

A. FOR THE STABILITY OF THE NUMERICAL SOLUTION **OBTAINED ABM METHOD**

This is clear from (6.4) and (6.5) that the functions $H_{n,1}(t), H_{n,2}(t), H_{n,3}(t), H_{n,4}(t), and H_{n,5}(t)$ Satisfies the Lipschitz condition,

Now,

$$\begin{aligned} \left\| H_{n,1}(t_k) - H_{n,1}(t_{k-1}) \right\|_{\infty} \\ &= \left\| -\beta S_n(t_k) \ I_n(t_k) - (\delta_1 + \gamma) S_n(t_k) \right. \\ &+ \beta S_n(t_{k-1}) \ I_n(t_{k-1}) + (\delta_1 + \gamma) S_n(t_{k-1}) \right\|_{\infty} \tag{6.17}$$

using (6.12) and (6.13) in (6.17) and from ref. (60)



FIGURE 2. Basic Reproduction number less than one $(R_0^{th} = 0.316235)$.



FIGURE 3. Basic Reproduction number greater than one ($R_0^{th} = 1.192941$).



FIGURE 4. Analyse the effect of fractional derivative (α) on infectious nodes.

for $k \to \infty$ as $||H_{n,1}(t_k) - H_{n,1}(t_{k-1})||_{\infty} \to 0$, therefore, from Theorem 3.3, the solution $S_n(t_k)$ is stable.

Similarly, we can obtain the stability condition for solutions $E_n(t_k)$, $I_n(t_k)$, $V_n(t_k)$ and $R_n(t_k)$.

VIII. RESULTS AND DISCUSSION

Conduct a series of numerical simulation experiments to study the impact of different factors on the spread of malware in WSN.



FIGURE 5. Analyse the effect of fractional derivative (α) on exposed nodes.



FIGURE 6. Analyse the effect of fractional derivative (α) on vaccinated nodes.



FIGURE 7. Analyse the effect of fractional derivative (α) on recovered nodes.

The proposed work is simulated using MATLAB on a 3-GHzIntel Xeon system running UBUNTU 19.2 LTS with 16-GBRAM. The results of simulation are illustrated in Figure 2 to Figure 7, for susceptible (S), Exposed(E), infectious (I), vaccinated and recovered (R) class with respect to time (t). The values of the various parameters considered for simulation are b = 1, $\beta = .02$, $\Upsilon = .2$, $\rho = .004$, $\eta = .07$, $\delta_1 = \delta_2 = \delta_3 = \delta_4 = \delta_5 = .1$ and taking the fractional derivative value $\alpha = .7$ In this analysis, it is

ascertainedthat initially infected nodes increase with time and achievethe maximum value. After that, it begins to disappear with timeand become zero. This shows that the system is asymptoticallystable in these conditions. The basic reproduction number incise 1 depicted in Figure 1, is0.316235 which is less than one, so malware disappears from the wireless sensor network. This result validates Theorems 5.1 and 5.2. When change some parametric value such as $\beta = .07$, $\rho =$.007, $\mu = .6$, and and other remain same. Then the value of $R_0^{th} = 1.192941$. Fig. 2 demonstrates that malware persists in the system continuously and in this case, the value of basic reproduction number (1.192941) is greater than 1. In this situation, malware will spreading the wireless sensor network., so malware continuously persist in the wireless sensor network. This result validates Theorems 5.3 and (5.4).

The impact of fractional derivative (α) on infectious nodes and exposed nodes are shown in Figure 3 and Figure 4. Shows that relationship between time and infectious nodes and exposed class nodes. It is evident from graph that when initially infectious class nodes and exposed class nodes achieve its maximum value for different value of α and after some time decline. For higher value of α the number of infectious nodes and exposed class nodes is less and for lower value of α the number of infectious nodes and exposed class nodes is more as compared to higher value of α .

Figure 5 shows the dynamic relationship between time and vaccinated class nodes for different value of α . It is clear from figure that when $\alpha = .095$ vaccination is very fast as compared to lesser value of α but after some time there is no impact of α on the vaccination. The link between time and recovered class nodes is shown in Figure 7, and it can be seen that as rises, recovery becomes quicker.

IX. CONCLUSION AND FUTURE WORK

The fractional-order derivatives are normally well suited for modelling since the determination about the derivative order offers one more degree of freedom, which results in a better fit to the real-time data with less inaccuracy than the integerorder model. Therefore, for controlling malware propagation and extending the lifespan of WSN, a fractional order SEIVR epidemic model is proposed. The proposed model offers a technique to quickly identify sensor node transitions from susceptible to infected state. Such detection provides a chance to further incorporate techniques for maintaining the sensor nodes and stabilizing WSN under various circumstances. The drawbacks of the earlier discussed model are overcome by an amended SEIRV model through the mechanism of vaccination. The computation of the basic reproduction number is one of the critical parameters that plays an important role in controlling malware transmission in the WSN. This value is computed. Additionally, it is found that endemic equilibrium and malware free equilibrium are locally and globally asymptotically stable, if the values of $R_0^{th} > 1$ and $R_0^{th} < 1$, respectively and this is supported by simulation results. These results are presented in Figure 2 and 3. Additionally, MAT-LAB(R2018a) simulations were run to test the validity of

the proposed model. Numerical solutions for the system have been found using the Adam-Bashforth-Moulton predictorcorrector method. The impact of various factors on malware propagation in WSN under various circumstances has been extensively analysed. Numerous results show that the proposed model offers a better defence mechanism against malware attacks. Further, fractional derivative models can be used to develop more effective control strategies for epidemics, including quarantine measures, and social distancing like policies depends on the type of network. There is a need for more research on how to use these models to optimize control strategies and minimize the impact of epidemics on WSN as well as online social network. In upcoming research work, we proposed our research in the field of variable order fractional derivative model through artificial intelligencebased approach.

REFERENCES

- S. Tang and W. Li, "QoS supporting and optimal energy allocation for a cluster based wireless sensor network," *Comput. Commun.*, vol. 29, nos. 13–14, pp. 2569–2577, Aug. 2006, doi: 10.1016/j.comcom.2006.02.007.
- [2] B. K. Mishra and N. Keshri, "Mathematical model on the transmission of worms in wireless sensor network," *Appl. Math. Model.*, vol. 37, no. 6, pp. 4103–4111, Mar. 2013, doi: 10.1016/j.apm.2012.09.025.
- [3] G. S. Rawat and K. Singh, "Joint beacon frequency and beacon transmission power adaptation for Internet of Vehicles," *Trans. Emerg. Telecommun. Technol.*, vol. 33, no. 10, Oct. 2022, doi: 10.1002/ett.4124.
- [4] H. Hu, Y. Han, H. Wang, M. Yao, and C. Wang, "Trust-aware secure routing protocol for wireless sensor networks," *ETRI J.*, vol. 43, no. 4, pp. 674–683, Aug. 2021, doi: 10.4218/etrij.2020-0147.
- [5] D. Arivudainambi, R. Pavithra, and P. Kalyani, "Cuckoo search algorithm for target coverage and sensor scheduling with adjustable sensing range in wireless sensor network," *J. Discretes Math. Sci. Cryptogr.*, vol. 24, no. 4, pp. 975–996, May 2021, doi: 10.1080/09720529.2020.1753301.
- [6] P. Bhambu and S. Kumar, "Target K-coverage problem in wireless sensor networks," J. Discrete Math. Sci. Cryptogr., vol. 23, no. 2, pp. 651–659, Feb. 2020, doi: 10.1080/09720529.2020.1729511.
- [7] A. Gambhir, A. Payal, and R. Arya, "Water cycle algorithm based optimized clustering protocol for wireless sensor network," *J. Interdiscipl. Math.*, vol. 23, no. 2, pp. 367–377, Feb. 2020, doi: 10.1080/09720502.2020.1731950.
- [8] B. Sandywell, "Monsters in cyberspace cyberphobia and cultural panic in the information age," *Inf., Commun. Soc.*, vol. 9, no. 1, pp. 39–61, Feb. 2006, doi: 10.1080/13691180500519407.
- [9] M. Gao, Y. Wang, Y. Wang, Y. Yao, P. Wang, Y. Sun, and J. Xiao, "Modeling and experimental verification of a fractional damping quad-stable energy harvesting system for use in wireless sensor networks," *Energy*, vol. 190, Jan. 2020, Art. no. 116301, doi: 10.1016/j.energy.2019.116301.
- [10] J. Kim, T. Kim, J. Noh, and S. Cho, "Fractional frequency reuse scheme for device to device communication underlaying cellular on wireless multimedia sensor networks," *Sensors*, vol. 18, no. 8, p. 2661, Aug. 2018, doi: 10.3390/s18082661.
- [11] A. G. M. Selvam, S. G. Winster, R. Janagaraj, and G. M. Jones, "Modeling worm proliferation in wireless sensor networks with discrete fractional order system," *Int. J. Recent Technol. Eng.*, vol. 8, no. 5, pp. 1815–1820, Jan. 2020, doi: 10.35940/ijrte.E4594.018520.
- [12] M. Liu, "Analysis for multi-coverage problem in wireless sensor networks," J. Softw., vol. 18, no. 1, p. 127, 2007, doi: 10.1360/jos180127.
- [13] Z. Vincze, R. Vida, and A. Vidacs, "Deploying multiple sinks in multi-hop wireless sensor networks," in *Proc. IEEE Int. Conf. Pervasive Services*, Jul. 2007, pp. 55–63, doi: 10.1109/PERSER.2007.4283889.
- [14] A. Singh, A. K. Awasthi, K. Singh, and P. K. Srivastava, "Modeling and analysis of worm propagation in wireless sensor networks," *Wireless Pers. Commun.*, vol. 98, no. 3, pp. 2535–2551, Feb. 2018, doi: 10.1007/s11277-017-4988-3.
- [15] I. Koca, "Analysis of rubella disease model with non-local and nonsingular fractional derivatives," *Int. J. Optim. Control, Theories Appl.*, vol. 8, no. 1, pp. 17–25, Dec. 2017, doi: 10.11121/ijocta.01.2018.00532.

- [16] A. Taghvaei, T. T. Georgiou, L. Norton, and A. Tannenbaum, "Fractional SIR epidemiological models," *Sci. Rep.*, vol. 10, no. 1, p. 20882, Nov. 2020, doi: 10.1038/s41598-020-77849-7.
- [17] C. Baishya, S. J. Achar, P. Veeresha, and D. G. Prakasha, "Dynamics of a fractional epidemiological model with disease infection in both the populations," *Chaos: Interdiscipl. J. Nonlinear Sci.*, vol. 31, no. 4, Apr. 2021, Art. no. 043130, doi: 10.1063/5.0028905.
- [18] R. K. Upadhyay, S. Kumari, and A. K. Misra, "Modeling the virus dynamics in computer network with SVEIR model and nonlinear incident rate," *J. Appl. Math. Comput.*, vol. 54, nos. 1–2, pp. 485–509, Jun. 2017, doi: 10.1007/s12190-016-1020-0.
- [19] G. Shrivastava, P. Kumar, R. P. Ojha, P. K. Srivastava, S. Mohan, and G. Srivastava, "Defensive modeling of fake news through online social networks," *IEEE Trans. Computat. Social Syst.*, vol. 7, no. 5, pp. 1159–1167, Oct. 2020, doi: 10.1109/TCSS.2020.3014135.
- [20] R. P. Ojha, P. K. Srivastava, and G. Sanyal, "Improving wireless sensor networks performance through epidemic model," *Int. J. Electron.*, vol. 106, no. 6, pp. 862–879, Jun. 2019, doi: 10.1080/00207217.2019.1570563.
- [21] S. Tang, "A modified Si epidemic model for combating virus spread in wireless sensor networks," *Int. J. Wireless Inf. Netw.*, vol. 18, no. 4, pp. 319–326, Dec. 2011, doi: 10.1007/s10776-011-0147-z.
- [22] X. Wang, Q. Li, and Y. Li, "EiSIRS: A formal model to analyze the dynamics of worm propagation in wireless sensor networks," *J. Combinat. Optim.*, vol. 20, no. 1, pp. 47–62, Jul. 2010, doi: 10.1007/s10878-008-9190-9.
- [23] N. Keshri and B. K. Mishra, "Two time-delay dynamic model on the transmission of malicious signals in wireless sensor network," *Chaos, Solitons Fractals*, vol. 68, pp. 151–158, Nov. 2014, doi: 10.1016/j.chaos.2014.08.006.
- [24] K. Diethelm, "The analysis of fractional differential equations: An application-oriented exposition using differential operators of Caputo type," Berlin, Germany, Tech. Rep., 2010.
- [25] I. Podlubny, "Fractional differential equations," New York, NY, USA, Tech. Rep., 1998.
- [26] A. A. Kilbas, H. M. Srivastava, and J. J. Trujillo, *Theory and Applications of Fractional Differential Equations*, 2006.
 [27] V. Srivastava and K. N. Rai, "A multi-term fractional diffusion
- [27] V. Srivastava and K. N. Rai, "A multi-term fractional diffusion equation for oxygen delivery through a capillary to tissues," *Math. Comput. Model.*, vol. 51, nos. 5–6, pp. 616–624, Mar. 2010, doi: 10.1016/j.mcm.2009.11.002.
- [28] V. Srivastava and K. N. Rai, "Approximate analytical solution of 3D fractional microscale heat equation using modified homotopy perturbation method," *Appl. Math. Sci.*, vol. 3, no. 32, pp. 1557–1565, 2009.
- [29] V. Srivastava, D. Tripathi, and O. A. Bég, "Numerical study of oxygen diffusion from capillary to tissues during hypoxia with external force effects," *J. Mech. Med. Biol.*, vol. 17, no. 2, Mar. 2017, Art. no. 1750027, doi: 10.1142/S0219519417500270.
- [30] L. Bolton, A. H. J. J. Cloot, S. W. Schoombie, and J. P. Slabbert, "A proposed fractional-order Gompertz model and its application to tumour growth data," *Math. Med. Biol., J. IMA*, vol. 32, no. 2, pp. 187–209, Jun. 2015.
- [31] C. Huang, H. Li, and J. Cao, "A novel strategy of bifurcation control for a delayed fractional predator-prey model," *Appl. Math. Comput.*, vol. 347, pp. 808–838, Apr. 2019, doi: 10.1016/j.amc.2018.11.031.
- [32] B. Ghanbari and D. Kumar, "Numerical solution of predator-prey model with Beddington–DeAngelis functional response and fractional derivatives with Mittag–Leffler kernel," *Chaos, Interdiscipl. J. Nonlinear Sci.*, vol. 29, no. 6, Jun. 2019, Art. no. 063103, doi: 10.1063/1.5094546.
- [33] C. Baishya, "Dynamics of fractional Holling type-II predator-prey model with prey refuge and additional food to predator," *J. Appl. Nonlinear Dyn.*, vol. 10, no. 2, pp. 315–328, Jun. 2021, doi: 10.5890/JAND.2021. 06.010.
- [34] R. P. Agarwal, O. Bazighifan, and M. A. Ragusa, "Nonlinear neutral delay differential equations of fourth-order: Oscillation of solutions," *Entropy*, vol. 23, no. 2, p. 129, Jan. 2021, doi: 10.3390/e23020129.
- [35] R. Khoshsiar Ghaziani, J. Alidousti, and A. B. Eshkaftaki, "Stability and dynamics of a fractional order Leslie–Gower prey–predator model," *Appl. Math. Model.*, vol. 40, no. 3, pp. 2075–2086, Feb. 2016, doi: 10.1016/j.apm.2015.09.014.
- [36] I. Petras, "Nonlinear physical science," London, U.K., Tech. Rep., 2011.
- [37] J. Huo, H. Zhao, and L. Zhu, "The effect of vaccines on backward bifurcation in a fractional order HIV model," *Nonlinear Anal., Real World Appl.*, vol. 26, pp. 289–305, Dec. 2015, doi: 10.1016/j.nonrwa.2015.05.014.
- [38] M. Caputo and M. Frabrizio, "A new definition of fractional derivative without singular kernel," *Prog. Fract. Differ. Appl.*, vol. 1, no. 2, pp. 73–85, 2015, doi: 10.12785/pfda/010201.

- [39] M. Ali Dokuyucu, E. Celik, H. Bulut, and H. Mehmet Baskonus, "Cancer treatment model with the Caputo–Fabrizio fractional derivative," *Eur. Phys. J. Plus*, vol. 133, no. 3, p. 92, Mar. 2018, doi: 10.1140/epjp/i2018-11950-y.
- [40] D. Baleanu, A. Jajarmi, H. Mohammadi, and S. Rezapour, "A new study on the mathematical modelling of human liver with Caputo–Fabrizio fractional derivative," *Chaos, Solitons Fractals*, vol. 134, May 2020, Art. no. 109705, doi: 10.1016/j.chaos.2020.109705.
- [41] A. Atangana and D. Baleanu, "New fractional derivatives with nonlocal and non-singular kernel: Theory and application to heat transfer model," *Thermal Sci.*, vol. 20, no. 2, pp. 763–769, 2016.
 [42] A. Atangana and I. Koca, "Chaos in a simple nonlinear system
- [42] A. Atangana and I. Koca, "Chaos in a simple nonlinear system with Atangana–Baleanu derivatives with fractional order," *Chaos, Solitons Fractals*, vol. 89, pp. 447–454, Aug. 2016, doi: 10.1016/j.chaos.2016.02.012.
- [43] M. Toufik and A. Atangana, "New numerical approximation of fractional derivative with non-local and non-singular kernel: Application to chaotic models," *Eur. Phys. J. Plus*, vol. 132, no. 10, p. 444, Oct. 2017, doi: 10.1140/epjp/i2017-11717-0.
- [44] D. G. Prakasha, P. Veeresha, and H. M. Baskonus, "Analysis of the dynamics of hepatitis e virus using the Atangana–Baleanu fractional derivative," *Eur. Phys. J. Plus*, vol. 134, no. 5, p. 241, May 2019, doi: 10.1140/epjp/i2019-12590-5.
- [45] B. Ghanbari and A. Atangana, "A new application of fractional Atangana–Baleanu derivatives: Designing ABC-fractional masks in image processing," *Phys. A, Stat. Mech. Appl.*, vol. 542, Mar. 2020, Art. no. 123516, doi: 10.1016/j.physa.2019.123516.
- [46] K. A. Abro and J. F. Gómez-Aguilar, "A comparison of heat and mass transfer on a Walter's-B fluid via Caputo–Fabrizio versus Atangana– Baleanu fractional derivatives using the fox-H function," *Eur. Phys. J. Plus*, vol. 134, no. 3, pp. 1–12, Mar. 2019, doi: 10.1140/EPJP/I2019-12507-4.
- [47] K. A. Abro, M. H. Laghari, and J. F. Gómez-Aguilar, "Application of Atangana–Baleanu fractional derivative to carbon nanotubes based non-newtonian nanofluid: Applications in nanotechnology," *J. Appl. Comput. Mech.*, vol. 6, no. 2, pp. 1260–1269, Dec. 2020, doi: 10.22055/JACM.2020.33461.2229.
- [48] J. F. Gómez-Aguilar, "Chaos in a nonlinear Bloch system with Atangana–Baleanu fractional derivatives," *Numer. Methods Partial Differ. Equ.*, vol. 34, no. 5, pp. 1716–1738, Sep. 2018, doi: 10.1002/NUM.22219.
- [49] J. F. Gómez-Aguilar, K. A. Abro, O. Kolebaje, and A. Yildirim, "Chaos in a calcium oscillation model via Atangana–Baleanu operator with strong memory," *Eur. Phys. J. Plus*, vol. 134, no. 4, pp. 1–9, Apr. 2019, doi: 10.1140/EPJP/I2019-12550-1.
- [50] V. F. Morales-Delgado, J. F. Gómez-Aguilar, K. Saad, and R. F. E. Jiménez, "Application of the Caputo–Fabrizio and Atangana–Baleanu fractional derivatives to mathematical model of cancer chemotherapy effect," *Math. Methods Appl. Sci.*, vol. 42, no. 4, pp. 1167–1193, Mar. 2019, doi: 10.1002/MMA.5421.
- [51] H. Tajadodi, A. Khan, J. F. Gómez-Aguilar, and H. Khan, "Optimal control problems with Atangana–Baleanu fractional derivative," *Optim. Control Appl. Methods*, vol. 42, no. 1, pp. 96–109, Jan. 2021, doi: 10.1002/OCA.2664.
- [52] A. Atangana and J. F. Gómez-Aguilar, "Fractional derivatives with noindex law property: Application to chaos and statistics," *Chaos, Solitons Fractals*, vol. 114, pp. 516–535, Sep. 2018, doi: 10.1016/j.chaos.2018.07.033.
- [53] S. T. M. Thabet, M. S. Abdo, K. Shah, and T. Abdeljawad, "Study of transmission dynamics of COVID-19 mathematical model under ABC fractional order derivative," *Results Phys.*, vol. 19, Dec. 2020, Art. no. 103507, doi: 10.1016/j.rinp.2020.103507.
- [54] A. González-Calderón, L. X. Vivas-Cruz, M. A. Taneco-Hernández, and J. F. Gómez-Aguilar, "Assessment of the performance of the hyperbolic-NILT method to solve fractional differential equations," *Math. Comput. Simul.*, vol. 206, pp. 375–390, Apr. 2023.
- [55] S. Kumar, V. Gupta, A. Kumar, and J. F. Gómez-Aguilar, "An accurate operational matrix method based on Lagrange polynomials for solving fractional-order pantograph delay and Riccati differential equations," *Phys. Scripta*, vol. 98, no. 4, Apr. 2023, Art. no. 044005.
- [56] P. Pandey and J. F. Gómez-Aguilar, "On solution of a class of nonlinear variable order fractional reaction-diffusion equation with Mittag-Leffler kernel," *Numer. Methods Partial Differ. Equ.*, vol. 37, no. 2, pp. 101–998, Mar. 2021.
- [57] S. Djennadi, N. Shawagfeh, M. S. Osman, J. F. Gómez-Aguilar, and O. A. Arqub, "The Tikhonov regularization method for the inverse source problem of time fractional heat equation in the view of ABC-fractional technique," *Phys. Scripta*, vol. 96, no. 9, Sep. 2021, Art. no. 094006.
- VOLUME 11, 2023

- [58] V. F. Morales-Delgado, M. A. Taneco-Hernández, C. Vargas-De-León, and J. F. Gómez-Aguilar, "Exact solutions to fractional pharmacokinetic models using multivariate Mittag–Leffler functions," *Chaos, Solitons Fractals*, vol. 168, pp. 113–164, Jan. 2023.
- [59] S. Paul, A. Mahata, S. Mukherjee, B. Roy, M. Salimi, and A. Ahmadian, "Study of fractional order SEIR epidemic model and effect of vaccination on the spread of COVID-19," *Int. J. Appl. Comput. Math.*, vol. 8, no. 5, p. 237, Oct. 2022, doi: 10.1007/s40819-022-01411-4.
- [60] A. Atangana and K. M. Owolabi, "New numerical approach for fractional differential equations," *Math. Model. Natural Phenomena*, vol. 13, no. 1, p. 3, 2018.
- [61] K. M. Owolabi and A. Atangana, Numerical Methods for Fractional Differentiation (Springer Series in Computational Mathematics), vol. 54, 2019.
- [62] G. Wu, L. Xie, H. Zhang, J. Wang, S. Shen, and S. Yu, "STSIR: An individual-group game-based model for disclosing virus spread in social Internet of Things," *J. Netw. Comput. Appl.*, vol. 214, May 2023, Art. no. 103608, doi: 10.1016/J.JNCA.2023.103608.
- [63] H. Zhang, S. Yang, G. Wu, S. Shen, and Q. Cao, "Steady-state availability evaluation for heterogeneous edge computing-enabled WSNs with malware infections," *Mobile Inf. Syst.*, vol. 2022, pp. 1–16, Apr. 2022, doi: 10.1155/2022/4743605.
- [64] X. Ye, S. Xie, and S. Shen, "SIR1R2: Characterizing malware propagation in WSNs with second immunization," *IEEE Access*, vol. 9, pp. 82083–82093, 2021, doi: 10.1109/ACCESS.2021.3086531.
- [65] S. Shen, H. Zhou, S. Feng, J. Liu, H. Zhang, and Q. Cao, "An epidemiology-based model for disclosing dynamics of malware propagation in heterogeneous and mobile WSNs," *IEEE Access*, vol. 8, pp. 43876–43887, 2020, doi: 10.1109/ACCESS.2020.2977966.
- [66] H. Zhou, S. Shen, and J. Liu, "Malware propagation model in wireless sensor networks under attack-defense confrontation," *Comput Commun*, vol. 162, pp. 51–58, Oct. 2020, doi: 10.1016/J.COMCOM.2020.08.009.
- [67] H. Zhang, S. Shen, Q. Cao, X. Wu, and S. Liu, "Modeling and analyzing malware diffusion in wireless sensor networks based on cellular automaton," *Int. J. Distrib. Sensor Netw.*, vol. 16, no. 11, Nov. 2020, Art. no. 155014772097294.
- [68] S. Shen, H. Zhou, S. Feng, J. Liu, and Q. Cao, "SNIRD: Disclosing rules of malware spread in heterogeneous wireless sensor networks," *IEEE Access*, vol. 7, pp. 92881–92892, 2019, doi: 10.1109/ACCESS.2019.2927220.
- [69] S. Shen, H. Zhou, S. Feng, L. Huang, J. Liu, S. Yu, and Q. Cao, "HSIRD: A model for characterizing dynamics of malware diffusion in heterogeneous WSNs," *J. Netw. Comput. Appl.*, vol. 146, Nov. 2019, Art. no. 102420, doi: 10.1016/J.JNCA.2019.102420.
- [70] Q. Kong. (2014). A Short Course in Ordinary Differential Equations. [Online]. Available: http://www.springer.com/series/223
- [71] Y. Li, Y. Chen, and I. Podlubny, "Stability of fractional-order nonlinear dynamic systems: Lyapunov direct method and generalized Mittag–Leffler stability," *Comput. Math. Appl.*, vol. 59, no. 5, pp. 1810–1821, Mar. 2010, doi: 10.1016/j.camwa.2009.08.019.
- [72] C. Vargas-De-León, "Volterra-type Lyapunov functions for fractionalorder epidemic systems," *Commun. Nonlinear Sci. Numer. Simul.*, vol. 24, nos. 1–3, pp. 75–85, Jul. 2015, doi: 10.1016/j.cnsns.2014.12.013.
- [73] M. Saeedian, M. Khalighi, N. Azimi-Tafreshi, G. R. Jafari, and M. Ausloos, "Memory effects on epidemic evolution: The susceptibleinfected-recovered epidemic model," *Phys. Rev. E, Stat. Phys. Plasmas Fluids Relat. Interdiscip. Top.*, vol. 95, no. 2, Feb. 2017, Art. no. 022409, doi: 10.1103/PhysRevE.95.022409.



VINEET SRIVASTAVA received the M.Sc. degree in mathematics from Banaras Hindu University (BHU), Varanasi, India, in 2003, and the Ph.D. degree in applied mathematics from the Indian Institute of Technology IIT (BHU) Varanasi, Varanasi, in 2009. He is currently an Assistant Professor with the Department of Mathematics, Rajkiya Engineering College Azamgarh (Affiliated to Dr.A. P. J. Abdul Kalam Technical University Lucknow), India. He has 13 years of teaching

and research experience along with Ph.D. guidance. He has several publications in various reputed international journals and international conferences. His research interests include mathematical modeling, dynamical systems, fractional order systems, and numerical techniques.



PRAMOD KUMAR SRIVASTAVA received the Ph.D. degree in mathematics from the University of Allahabad, Allahabad, India. He is currently the Dean (student welfare) of the Rajkiya Engineering College Azamgarh (Affiliated to Dr. A. P. J. Abdul Kalam Technical University Lucknow), India. He has several publications in various reputed international journals and international conferences. He has also published multiple books. His research interests

include wireless sensor networks, mathematical modeling, and social networks. He has more than 15 years of teaching and research experience along with Ph.D. guidance. He has successfully guided two Ph.D. students and is currently guiding a few more. Excellence in learning has been his motto since he was a student and the proof of this was the gold medal awarded to him for securing the highest marks in his post-graduation. He has applied this motto in his teaching too, and being well known in his genre, he has been invited to give guest lectures at reputed institutes and international conferences. He has been a member of organizing committees and technical committees of many international conferences. He is renowned not only for the intellectual treasure he possesses, but also for his benevolence and altruism.



JYOTI MISHRA received the B.Sc. degree from the Agrasen Kanya P. G. College (affiliated from Mahatma Gandhi Kashi Vidyapeeth), Varanasi, India, in 2019, and the M.Sc. degree from Banaras Hindu University, Varanasi, in 2021. She is currently pursuing the Ph.D. degree with Abdul Kalam Technical University, Lucknow, India, under Homi Bhabha Teaching Assistant Fellowship.



RUDRA PRATAP OJHA received the B.Tech. degree from UPTU Lucknow, India, the M.Tech. degree from the Motilal Nehru National Institute of Technology (MNNIT) Allahabad, India, and the Ph.D. degree from the National Institute of Technology (NIT) Durgapur, India. Currently, he is affiliated with the G. L. Bajaj Institute of Technology and Management, Greater Noida, India, where he is also a Professor with the Computer Science Engineering Department. He is also an

academician and a researcher in computer science engineering. He is also the Head of Planning and the Head of Internal Quality Assurance Cell (IQAC). He has published over 20 papers in various international and national journals and conferences. In addition, he serves as a reviewer for various reputed journals. His research interests include wireless sensor networks, real-time systems, mathematical modeling, and simulation.



PURNENDU SHEKHAR PANDEY (Senior Member, IEEE) received the Ph.D. degree from the Indian Institute of Technology (Indian School of Mines) Dhanbad, India. Currently, he is with the Electronics and Communication Engineering Department, GL Bajaj Institute of Technology and Management, Greater Noida, India, as an Assistant Professor. He has over one decades of experience in academics and research. He has published more than 55 high-quality research papers in national

and international journals and conferences. He has also authored a text book titled *Power Line Carrier Communication and Arduino Based Automation*. He was also a JRF under the Indian Space Research Organization (ISRO) sponsored Project (Sanction Nostro/RES/3/775/18-19) with the Indian Institute of Technology (Indian School of Mines) Dhanbad. He holds ten patents

published and one granted patent. He is a member of SPIE. He is also a reviewer in reputed journals. His current research interests include optical sensors, nano and bio photonics, photonic and plasmonic devices, and wireless sensor networks, biomedical engineering, the IoT, and artificial intelligence.



RADHE SHYAM DWIVEDI received the B.Tech. degree from Dr. A. P. J. Abdul Kalam Technical University Lucknow (UPTU Lucknow), Lucknow, India, in 2008, and the M.Tech. degree from the Birla Institute of Technology Mesra, Ranchi, in 2015. He is currently pursuing the Ph.D. degree with the Motilal Nehra National Institute of Technology. He is also an Assistant Professor with the Department of Applied Sciences and Humanities, Rajkiya Engineering College Azam-

garh (affiliated to Dr. A. P. J. Abdul Kalam Technical University Lucknow), India, for the last five years. His research interests include mathematical modeling, sliding mode control, and PWM control power converters.



LORENZO CARNEVALE (Member, IEEE) received the Ph.D. degree from the University of Messina, in May 2020. He is currently an Assistant Professor with the University of Messina. He works in artificial intelligence and distributed systems with applications in the healthcare and robotics sectors. He has led technical tasks for European and national projects, i.e., H2020 "FLI-WARE," FISR "The Re-functionalization of the Contemporary," and HE TEMA. He is the coau-

thor of over 35 scientific publications and patents in *Distributed Systems*, *Computer Networks*, *Artificial Intelligence Systems*, and *Robotics*. He has been the PC Chair of DistInSys, MrICHE, AI4Health, and UCC 2023. He is a member of the Working Group 12.9—Computational Intelligence of the International Federation for Information Processing (IFIP) and the Coordinator of the CINI InfoLife, University of Messina.



ANTONINO GALLETTA (Member, IEEE) received the B.Sc. and M.Sc. degrees (Hons.) in computer engineering from the University of Messina, and the Ph.D. degree from the University of Reggio Calabria, Italy, in 2020. He was a Senior Software Developer and a Team Leader with the University of Messina, from 2013 to 2016, where he is currently an Assistant Professor. He is also the PI of "InstradaME" an Italian Project funded by the Ministry of Interior. In 2019, he was the PI of

NAME (Helsinki Noise and Air quality Monitoring systEm) a project funded by the European Pre-Commercial Procurement Project "Select for Cities" (Grant Agreement No 688196). His research interests include the security of cloud/edge/IoT technologies for smart cities and eHealth solutions, including big data management and blockchain. Due to his contribution to the research, he has been selected among 200 top young researchers in mathematics and computer science to participate in the prestigious "Heidelberg Laureate Forum" held in Heidelberg, in September2019. He has been the winner of two editions of the precious "Leonardo Innovation Award," in 2018 (first placed) and 2017 (second placed). In 2019, he was one of the winners of the FIWARE Genoa Hackathon. In December 2018, he won the Vienna FIWARE challenge and in April, during the "International Conference on Geoinformatics and Data Analysis" (ICGDA, Prague April), he won the "Best Presentation Award." He was the guest editor of several SI on Q1 journals. He is also an Associate Editor of SusCom (Elsevier); one of the members of the Technical Committee of the Computer Communication (Elsevier), and a reviewer of more than 15 respected journals. He is the Chair of several international conferences and the author of more than 60 manuscripts.

• • •

Open Access funding provided by 'Università degli Studi di Messina' within the CRUI CARE Agreement